DATED 2022

THE SECRETARY OF STATE FOR (1)
WORK AND PENSIONS

AND

VINCI CONSTRUCTION UK LIMITED (2)

CONTRACT FOR THE APPOINTMENT
OF A CONTRACTOR RELATING TO
THE CONSTRUCTION WORKS AND
ASSOCIATED SERVICES IN ENGLAND
(SOUTH) AND INCORPORATING THE
NEC4 ENGINEERING AND
CONSTRUCTION CONTRACT
LOT C

THIS CONTRACT is made on

2022 **BETWEEN**:

(1) **THE SECRETARY OF STATE FOR WORK AND PENSIONS** acting as part of the Crown (the "**Client**") (which expression shall include any successors in title and assigns); and

(2) Vinci Construction UK Limited (company number VINCI Construction UK Limited whose registered office is at Astral House, Imperial Way, Watford, Hertfordshire, WD24 4WW the "Contractor"),

together, the Parties.

WHEREAS:

- (A) The Client wishes to enter into a contract with the Contract on the terms set out in this Contract for the carrying out and completion of the Works.
- (B) The Client wishes to retain the Contractor to carry out certain additional works at the Sites which may be notified to the Contractor from time to time during the Delivery Term.
- (C) In the event that the Client requires the Contractor to carry out work in relation to any part of the Project, such works will be carried out under a Task Order.
- (D) The Client and Contractor accept and agree that this Contract does not guarantee that any work will be given by the Client to the Contractor or oblige the Client to instruct the Contractor to carry out any particular scope of work.

In consideration of £1 paid by each party to the other, receipt of which each recipient party acknowledges **IT IS HEREBY AGREED** as follows:

1 <u>Introduction</u>

- 1.1 In this Contract the following words and expressions have the following meanings unless the context otherwise requires:
 - "Business Days" means a day (other than a Saturday or Sunday or Bank Holiday) on which banks are open for domestic business in the City of London;
 - "Capital Works Fee Percentage" means the fee percentage in relation to capital works as set out in the Capital Works Tender Rates and Prices;
 - "Capital Works Price List" means the price list relating to capital works attached at Appendix 5, incorporating the Capital Works Tender Rates and Prices, where applicable and setting out the Fee Percentage;
 - "Capital Works Tender Rates and Prices" means the rates and prices submitted by the Contractor in relation to capital works for the Project, which shall be priced with reference to the Estates Expansion Programme Schedule of Rates referred to in the ITT, and taking into account but not exceeding the applicable CCS Framework Rates and Prices;
 - "CDM Regulations" means the CDM Regulations 2015;
 - "CCS Framework Rates and Prices" means those rates and prices that apply to the Crown Commercial Service framework reference RM 6088, Lot 2.2
 - "Confidential Information" means all confidential information (however recorded or preserved) disclosed by the Client or the Contractor to the other party whether before

or after the date of this Contract in connection with the Works, including but not limited to:

- (a) the terms of this Contract;
- (b) the fact that discussions and negotiations are taking place concerning the Works and/or the Site and the status of those discussions and negotiations;
- (c) any information that would be regarded as confidential by a reasonable businessperson relating to:
 - (i) the business, affairs, customers, clients, suppliers, or plans, intentions, or market opportunities of the Client and the Contractor; and
 - (ii) the operations, processes, product information, know-how, designs, trade secrets or software of the opportunities of the Client or the Contractor; and
 - (iii) any information developed by the parties in the course of carrying out this Contract.

"Contract Data" means that contract data as applicable to the relevant Works and as set out at Appendix 2;

"Contract Month" means each calendar month during the Delivery Term.

"Critical Failure Event" a failure to provide any Works in accordance with this Contract and/or any relevant Task Order and the Client considers that such failure results in or may result in:

- (a) imminent risk of danger to persons.
- (b) imminent risk of full or partial loss of a Task Site; and/or
- (c) a material adverse effect upon the Client's reputation and/or business within the relevant Task Site (where applicable).

"Client's Policies" means the policies and directives or other regulations of the Client from time-to-time copies of which are provided and updated by the Client from time to time;

"Client's Schedules" means the following documents which are included at Appendix 11 within this document:

(a) Part A: Appendix G (Collateral Warranty); (b)

Part B: Appendix K (TUPE).

"Contractor's Tender Submission" means the Contractor's tender submitted to the Client for the Works attached at Appendix 13;

"CWAS Framework" means the CCS Framework Reference RM6088 in relation to "Construction Works and Associated Services;

"Delivery Term" means the period of two years and six months from the date of this contract as may be extended in accordance with the terms of this Contract and subject to early termination pursuant to the relevant Terms and Conditions.

"Earn Back Points" means the earn back points set out in column L of the table set out in Part B of Appendix 8 of this Contract and which shall accrue to the Contractor in where it exceeds the relevant Performance Level in a Service Period;

"Fee Percentage" means the Capital Works Fee Percentage and/or the Life Cycle Fee Percentage (as appropriate);

"ITT" means the invitation to tender issued by the Client as part of the procurement process relating to the CWAS Framework pursuant to which the Contractor was selected to enter into this Delivery Agreement;

"Law" is any law, statute, subordinate legislation within the meaning of section 21(1) of the Interpretation Act 1978, bye-law, enforceable right within the meaning of section 2 of the European Communities Act 1972, regulation, order, mandatory guidance or code of practice, judgement or a relevant court of law, or directives of any regulatory body with which the Contractor is bound to comply;

"Life Cycle Fee Percentage" means the fee percentage in relation to life cycle works as set out in the Life Cycle Tender Rates and Prices.

"Life Cycle Price List" means the price list relating to life cycle works attached at Appendix 5, incorporating the Life Cycle Tender Rates and Prices, where applicable and setting out the Fee Percentage.

"Life Cycle Tender Rates and Prices" means the rates and prices submitted by the Contractor in relation to life cycle works for the Project, which shall be priced with reference to the relevant PSA Schedules of Rates referred to in the ITT, and taking into account but not exceeding the applicable CCS Framework Rates and Prices;

"Management Services" means those services set out in Appendix 4;

"Necessary Consents" means are all permits, licences, permissions, consents, approvals, certificates and authorisations (whether statutory or otherwise) which are required for the performance of any of the Contractor's obligations under this Contract or any Task Order, whether required in order to comply with Law, Statutory Requirements or as a result of the rights of any third party;

"Notifiable Event" means:

- (a) there is an adverse change in the financial circumstances of the Contractor, the Contractor becomes aware of an adverse change in the financial circumstances of the Contractor has (or anticipates that it may have) insufficient funding to adequately resource its obligations under this Contract;
- (b) the Contractor becomes aware of any circumstances relating to the Contractor or any subcontractor which shall or may bring into disrepute and/or diminish the trust that the public places in the Client and/or the carrying out of the Works;

- (c) the Contractor is under investigation and/or subject to regulatory enforcement by any relevant Client and such investigation and/or enforcement shall or may be material to the Contractor's ability to provide the Works;
- (d) in the event that the Contractor commits a material default of this Contract and/or any Task Order that is capable of remedy;
- (e) the occurrence or likelihood of the occurrence of a Critical Failure Event;
- (f) the occurrence or likelihood of the occurrence of a Repeat Failure;
- (g) where the Contractor has been selected to take part in the Pricing Process (as defined in Appendix 1) and on three consecutive occasions:
 - (i) the Contractor has failed to submit a Task Order Proposal in accordance with Part B of Appendix 1; and/or
 - (ii) the Client is not able to approve the Contractor's Task Order Proposal in accordance with paragraph 2.2 of Part B of Appendix 1; and/or
- (h) where the Contractor receives a RED Score under Appendix 8; and/or
- (i) the Contractor fails to provide a Parent Company Guarantee in accordance with clause 23 of this Contract:

"NUTS Codes" means the Primary NUTS Codes and Secondary NUTS Codes;
"Objectives" means:

- (a) to deliver high quality Works;
- (b) to minimise disruption to any occupants of the Task Site;
- (c) to act in a collaborative manner and in the spirit of mutual trust and respect;
- (d) to adopt and demonstrate open book accounting techniques;
- (e) to achieve value for money and comply with Appendix 4 of this Contract;
- (f) to adopt innovative and sustainable working practices;
- (g) to promote practices that encourage a positive contribution to social value; and(h) to achieve continuous improvement.

"Other Consultants" means any other consultants appointed by the Client in connection with the Project and shall include those other consultants' employees, agents and sub-consultants;

"Parent Company" means the parent company nominated by the Contractor and approved by the Client;

"Parent Company Guarantee" means the form of guarantee attached at Appendix 9 with such amendments as the Client and the Contractor may agree, both acting reasonably;

"Performance Bond" means the form of performance bond attached at Appendix 10 with such amendments as the Client and the Contractor may agree, both acting reasonably;

"Price List" means the either the Capital Works Price List and/or the Life Cycle Price List (as appropriate);

"Prices" means the prices set out in any Task Order;

"Primary NUTS Codes" means the NUTS Codes identified as such in Appendix 3;

"Performance Failure" means a failure by the Contractor to meet the relevant Performance Level for a Performance KPI;

"Performance Failure Point" means the points that are set out in column P of the table in Part B of Appendix 8 of this Contract and which shall accrue to the Contractor in the event of any Performance Failure in a Service Period;

"Performance KPI" means a key performance indicator as set out at Part B of Appendix 8 and as may be updated from time to time in accordance with paragraph 5.1 of Part A of Appendix 8;

"Performance Level" the levels that are set out in Appendix 8 of this Contract; "Performance Monitoring Report" means the monthly performance report prepared by the Contractor in accordance paragraph 4.1 of Part A of Appendix 8;

"Performance Review Meeting" the regular meeting between the Contractor and the Client to discuss the Performance Monitoring Report, as set out in clause 5.3;

"Pipeline Programme" means the programme prepared by or on behalf of the Client for the Project as a whole, as updated from time to time;

"Project" means the Works comprising new building works and/or works of fit out, repair, replacement and/or refurbishment, including small works, minor works, intermediate works and/or life cycle works and related operations being undertaken at an office, shop, bureau, kiosk or other permanent or temporary building or unit used for the delivery of over-the-counter, telephone or consultant room style public services and advice and/or of commercial office and retail character envisaged under the CWAS Framework at the Sites which are carried out pursuant to Task Orders issued in accordance with Scope of Requirements;

"Project Participants" means the Client, the Contractor and (as relevant) the Delivery Contractors (as identified in Appendix 1), Other Consultants and any other consultant, sub-contractor and supplier appointed by the Client or the Contractor in relation to the Works and any obligations under this Contract;

"Project Risk Register" means the risk register created and updated in accordance with Part B of Appendix 4;

"**Project Team**" means the Client, the Contractor, and any other consultant, subcontractor and supplier appointed by the Client or the Contractor in relation to the Works and any obligations under this Contract;

"**Proposed Task**" means works comprising any element of the Project at any Task Site:

"Rates and Prices" means the rates and prices set out in the Price List and where applicable calculated in accordance with Appendix 5;

"Region" means England South;

"Required Insurances" means those insurances required for any Proposed Task and as further set out in the Task Order Proposal;

"Scope of Requirements" means the Client's scope document setting out the scope of requirements for the Works attached at Appendix 12;

"Senior Representatives" means at the Employer: REDACTED Associate Commercial Specialist, Estates Category; For the Contractor, REDACTED

"Service Period" means the service period for the relevant Performance KPI as set out in column G of the table set out in Part B of Appendix 8.

"Service Quarter" means the first three months and each subsequent three-month period (or reduced period immediately prior to the end of this Contract) which shall be defined as Jan – Mar inclusive, Apr – Jun inclusive, Jul – Sep inclusive and Oct – Dec inclusive in a calendar year;

"Short Schedule of Cost Components" or "SSCC" means the short schedule of cost components which forms part of the Terms and Conditions;

"SSCC Tender Rates and Prices" means any rates and prices in relation to the Short Schedule of Cost Components submitted by the Contractor as part of the Capital Works Tender Rates and Prices and/or the Life Cycle Tender Rates and Prices (as appropriate);

"Sites" means the relevant sites in the Region as further identified at Appendix 3 and any further sites instructed by the Client to the Contractor;

"Statutory Requirements" means:

- (a) any statute, regulation or subordinate legislation, the common law and any treaty or European Union law which relates to the protection of the environment or health and safety whether administrative, civil or criminal;
- (b) any statute, statutory instrument, regulation, rule or order made under any statute or directive having the force of law which affects the Works or the performance of any of the obligations under this Contract and/or any Task Order:
- (c) any regulation or byelaw of any local authority, statutory undertaker or similar body or organisation which has any jurisdiction with regard to the Works or with whose systems the Works are, or are to be, connected, including any statutory provisions and any decision of a relevant authority under them which controls the right to develop any of the Sites; and

(d) the consents of all persons who have rights or interests in or over the Works or the Sites who by the lawful exercise of their powers in the absence of such consents could delay or disrupt the carrying out or progress of the Works or the use or enjoyment of any of the Sites;

"Secondary NUTS Codes" means the NUTS Codes identified as such in Appendix 3;

"Task Order" means an appointment on the terms of which the Contractor will carry out works and/or services at the Task Site(s) and the terms of this Contract, the Terms and Conditions and the Contract Data shall be incorporated into the Task Order;

"Task Order Proposal" means a proposal issued by the Contractor to the Client following issue by the Client of a Project Instruction in accordance with Appendix 1 and in the format set out in Annex 1 of Appendix 1;

"Task Site(s)" means the site(s), being any one or more of the Sites (in the event that Works at multiple Sites are bundled into the same Task Order), to which the Task Order relates:

"Tender Rates and Prices" means the Capital Works Tender Rates and Prices, Life Cycle Tender Rates and Prices and/or the SSCC Tender Rates and Prices (as appropriate);

"**Terms and Conditions**" means the NEC4 Engineering and Construction Contract (Option A) and incorporating:

- (a) the schedule of amendments issued by the Crown Commercial Service under the framework agreement reference RM6088; and
- (b) those amendments set out in Appendix 1;

"Transparency Information" means the content of this Contract and any Task Order in their entirety but excluding the Contractor's Confidential Information, and any other information that would be exempt from disclosure:

- (c) under the CWAS Framework; and/or
- (d) under the FOIA and/or the EIR; and

"Works" means the works and/or services subsequently referred to in any Task Order (to include any works procured on a design and build basis).

- 1.2 Words importing the singular only shall also include the plural and vice versa, and where the context requires, words importing persons shall include firms and corporations.
- 1.3 The headings to the clauses of this Contract shall not affect its interpretation.
- 1.4 In this Contract, save as where set out in this Contract, words and expressions shall have the same meanings as are assigned to them in the Terms and Conditions.
- 1.5 Reference in this Contract to any Act of Parliament, statute or statutory instrument shall include and refer to any statutory amendment thereto from time to time and for the time

- being in force and any successor to such Act of Parliament, statute and statutory instrument.
- 1.6 Defined terms set out in the NEC4 Engineering and Construction Contract shall have effect in this Contract save where expressly set out.

2 Conditions

- 2.1 The terms and conditions of the NEC4 Engineering and Construction Contract Option A have effect as modified by this Contract, the Terms and Conditions and the Appendices.
- 2.2 Where a reference is made to a clause of the NEC4 Engineering and Construction Contract Option A (June 2017 Edition) contract it shall be read and construed to be a reference to that clause as amended by this Contract, the Terms and Conditions and Appendix 1.
- 2.3 The Client has appointed the Contractor to carry out certain services under a Task Order for a Site (a "Services Task Order"). Once a Task Order is instructed in relation to the works that the Services Task Order relates to (a "Works Task Order"), the Services Task Order shall be subsumed into that Works Task Order and any monies paid to the Contractor in respect of the Services Task Order shall be treated as payments on account in respect of the relevant Works Task Order. The Contractor agrees that it does not have any entitlement to any extension of time, compensation and/or any relief whatsoever under or pursuant to the Works Task Order in relation to any of the services carried out pursuant to the Services Task Order and the Contractor waives its entitlement to any extension of time, compensation and/or any relief whatsoever in relation to the Services Task Order once the Works Task Order has been instructed in accordance with this Contract.
- 2.4 Where the Client appoints the Contractor in relation to the Works which are the subject of a Services Task Order, the Parties agrees that the provisions of Part A of Appendix 1 shall not apply to the appointment of the Contractor in relation to the relevant Works Task Order.
- 2.5 In all matters in connection with this Contract, the Contractor and the Client shall act in good faith with due regard to the provisions of this Contract and the delivery of the Services and shall act reasonably and fairly with honest intention and without any intention to act maliciously or to defraud.

3 Delivery of Works

- 3.1 The Contractor shall carry out and complete the Works (to include any Works instructed by way of a Task Order) in accordance with the Terms and Conditions.
- 3.2 The Parties' obligations in relation to the identification, approval and delivery of the Works are contained in Appendix 1.
- 3.3 All Task Order Proposals shall by proposed by the Contractor in accordance with the provisions of Appendix 1.
- 3.4 Where a Task Order Proposal has been approved by the Client pursuant to Appendix 1, then the Client and the Contractor shall enter into a Task Order in respect of those Works on the basis of the Terms and Conditions.

3.5 The Client has no obligation to issue Task Orders to the Contractor and reserves the right to have any Works carried out or supplied by other contractors. The Client makes no representations regarding the level of Works it may appoint the Contractor to carry

out during the Delivery Term and does not guarantee minimum or maximum amounts of work under a particular Task Order or in total and no form of exclusivity is granted by the Client to the Contractor in relation to the Works.

4 Contractor's Duties

- 4.1 The Contractor shall perform the Works in accordance with each Task Order (as relevant), all Law and good industry practice having at all times due regard and so far as is reasonably practical to comply with any Pipeline Programme as may be issued by the Client to the Contractor from time to time.
- 4.2 The Contractor shall support the Client to ensure that all applicable parties comply with all Statutory Requirements and Necessary Consents in relation to health and safety across the Sites.
- 4.3 The Contractor shall obtain and maintain all relevant Necessary Consents required in order to carry out and complete the Works.
- 4.4 The Contractor shall carry out the Management Services in accordance with the provisions of Appendix 4.
- 4.5 Without prejudice to any other or related obligations in a Task Order, the Contractor shall, in complying with this Contract, any Task Order and its performance of the Works, act in a way which promotes the protection of the environment, sustainable construction and development and energy efficiency.
- 4.6 Without prejudice to any other or related obligations in a Task Order, the Contractor shall, in complying with this Contract, any Task Order and its performance of the Works, comply with the Statutory Requirements and Necessary Consents.
- 4.7 The Contractor shall comply with Appendix 8.

5 Notifiable Event

- 5.1 In the event that a Notifiable Event occurs, the Contractor shall notify the Client of the Notifiable Event as soon as practicable but in any event within three (3) Business Days of becoming aware of the Notifiable Event, detailing the actual or anticipated effect of the Notifiable Event.
- 5.2 If:
 - 5.2.1 the Contractor notifies the Client pursuant to Clause 5.1 that a Notifiable Event has occurred; or
 - 5.2.2 the Client notifies the Contractor that it considers that a Notifiable Event has occurred (setting out sufficient detail so that it is reasonably clear what the Contractor has to rectify),

then, without prejudice to any other right or remedy of the Client howsoever arising under this Contract and/or any relevant Task Order:

5.2.3 at the Client's option, give the Contractor the opportunity (at the Contractor's expense) to remedy the Notifiable Event together with any damage resulting from such Notifiable Event (where such Notifiable Event is capable of remedy) or to carry out any other necessary work to ensure

that the terms of this Contract and/or any relevant Task Order are fulfilled, in accordance with the Client's instructions;

5.2.4 where:

- the option to remedy has been provided but the Contractor failed to remedy the Notifiable Event within a reasonable period specified by the Client; or
- (ii) where the Notifiable Event is incapable of remedy, the Client may:
- (iii) step-in itself to carry out, or arrange for a third party to carry out, at the Contractor's expense, all or part of this Contract and/or any relevant Task Order:
- (iv) suspend (for any period) all or part of this Contract and/or any relevant Task Order; or
- (v) terminate all or part of this Contract and/or any relevant Task Order (and such termination shall be deemed to be an R11 termination event under the relevant Task Order).
- 5.3 Where the Client exercises any of its rights under Clause 5.2.4:
 - 5.3.1 the Client shall have the right to charge the Contractor for and the Contractor shall on demand pay any costs and expenses reasonably incurred by the Client (including any reasonable administration costs) in respect of:
 - (i) managing and implementing the step-in, suspension and/or termination (as the case may be);
 - (ii) the carrying out of all or part of this Contract and/or any relevant Task Order by the Client and/or a third party; and
 - (iii) any additional costs paid or allowable by the Client to a third party in relation to the carrying out of all or part of this Contract and/or any relevant Task Order by such third party;
 - 5.3.2 the Contractor shall not be obliged to provide the relevant Works to the extent that they are subject to any rights exercised by the Client under Clause 5.2.4; and
 - 5.3.3 any event in relation to the Client exercising its rights under Clause 5.2 shall not be deemed to be a compensation event for the purposes of any Task Order and the Contractor shall not be entitled to any addition to any Prices or additional costs and/or an extension of time under any Task Order:

6 <u>Limitation</u>

6.1 Notwithstanding that the Task Order may be executed under hand, either party may bring a claim, action or proceedings against the other after six years from the date of completion of Works provided that no claim, action or proceedings may be issued or brought against the Contractor after twelve years from the date of completion of the relevant Works unless such claim, action or proceedings were notified in writing to the Contractor prior to such date.

7 Extending this Contract

7.1 The Client can extend this Contract by giving no less than 1 month written notice prior to the expiry of the Delivery Term. More than one notice can be issued provided that this Contract is not extended by more than 6 months.

8 Senior Representatives

- 8.1 The Contractor shall appoint the Contractor's Senior Representative from the date of this Contract and throughout the Delivery Term the Contractor's Senior Representative shall be the Client's main point of contact in relation to the Contractor's engagement under any Task Order and in connection with all matters arising under or in connection with this Contract.
- 8.2 The Contractor's Senior Representative shall not be replaced without the Client's consent to the replacement in writing. The Client shall be entitled to withhold its consent where the proposed replacement is not of a similar level, qualification and with equivalent experience as his predecessor and/or for any other reasonable reason.

9 Performance review and meetings

- 9.1 The Contractor shall attend meetings no more than once every Service Quarter with the Client and the Delivery Contractors (as defined in Appendix 1) in order to share best practice and improve and development the performance of this Contract as a whole as and when reasonably requested to do so by the Client.
- 9.2 The Client and Contractor shall attend a meeting on the first anniversary of the Delivery Term (the "Annual Meeting") and at this meeting the Contractor may request, together with such supporting information as the Client may require, the selection of any Secondary NUTS Codes (where applicable). Where the Client agrees to such request, it shall take that into account when applying any subsequent Rotational Procedure.
- 9.3 At the Annual Meeting, the Contractor and the Client shall review the Pipeline Programme and any other relevant information to enable the Contractor to provide recommendations on bundling of certain works and/or packages in order to benefit from economies of scale, to reduce the number of individual Task Orders and/or to minimise disruption to the Client's core business.
- 9.4 The Contractor's Senior Representative shall attend all meetings required by this clause at no cost to the Client and shall at all times ensure that it is in possession of up to date information regarding the Contractor's performance on Task Orders.

10 <u>Termination</u>

- 10.1 The Client shall be entitled to suspend or terminate this Contract (with or without terminating any Task Orders let hereunder pursuant to its terms and the decision whether to so terminate both Task Orders and this Contract shall be at the Client's sole discretion) at any time by giving not less than 10 Business Days' prior written notice to the Contractor.
- 10.2 Termination shall be without prejudice to any accrued rights or remedies or any continuing obligations of either party and without prejudice to the continuation of any Task Orders that are not co-terminated with this Contract.
- 10.3 Notwithstanding any provision of this Contract or the terms of any Task Order the Contractor shall not be entitled to claim any loss of profit, loss of contracts, loss of opportunity or other indirect and/or consequential losses upon termination of its employment under this Contract or any Task Order.
- 10.4 The Contractor's obligations under clauses 5.3, 6 and 11 25 (inclusive) shall survive any termination of this Contract and/or of any Task Order.

11 <u>Disclosure of Information</u>

Freedom of Information

- 11.1 Subject to Clauses 11.4, 11.5, 11.8, 11.9 and 11.11 the Contractor:
 - 11.1.1 shall treat in confidence all Confidential Information it receives from the other;
 - 11.1.2 shall not disclose any of that Confidential Information to any third party without the prior written consent of the other party, which consent shall not unreasonably be withheld, except that the Contractor may disclose Confidential Information in confidence, without prior consent, to such persons and to such extent as may be necessary for the performance of the Contract;
 - 11.1.3 shall not use any of that Confidential Information otherwise than for the purpose of the Contract; and
 - 11.1.4 shall not copy any of that Confidential Information except to the extent necessary for the purpose of exercising its rights of use and disclosure under the Contract.
- 11.2 The Contractor shall take all reasonable precautions necessary to ensure that all Confidential Information disclosed to the Contractor by or on behalf of the Client under or in connection with the Contract:
 - 11.2.1 is disclosed to its employees and subcontractors, only to the extent necessary for the performance of the Contract; and
 - 11.2.2 is treated in confidence by them and not disclosed except with prior written consent or used otherwise than for the purpose of performing work or having work performed for the Client under the Contract.

- 11.3 The Contractor shall ensure that his employees are aware of his arrangements for discharging the obligations at Clauses 11.1 and 11.2 before they receive Confidential Information and take such steps as may be reasonably practical to enforce such arrangements.
- 11.4 Clauses 11.1 and 11.2 shall not apply to any Confidential Information to the extent that either party:
 - 11.4.1 exercises rights of use or disclosure granted otherwise than in consequence of, or under, the Contract;
 - 11.4.2 has the right to use or disclose the Confidential Information in accordance with other conditions of the Contract; or

11.4.3 can show:

- (i) that the Confidential Information was or has become published or publicly available for use otherwise than in breach of any provision of the Contract or any other agreement between the parties;
- (ii) that the Confidential Information was already known to it (without restrictions on disclosure or use) prior to it receiving it under or in connection with the Contract:
- (iii) that the Confidential Information was received without restriction on further disclosure from a third party who lawfully acquired it and who is himself under no obligation restricting its disclosure; or
- from its records that the same information was derived independently of that received under or in connection with the Contract,

provided the relationship to any other Confidential Information is not revealed.

11.5 Neither party shall be in breach of this Clause where it can show that any disclosure of Confidential Information was made solely and to the extent necessary to comply with a statutory, judicial or parliamentary obligation. Where such a disclosure is made, the party making the disclosure shall ensure that the recipient of the Confidential Information is made aware of and asked to respect its confidentiality. Such disclosure shall in no way diminish the obligations of the parties under this Clause.

11.6 The Contractor:

- 11.6.1 transfers to the Client all requests for information that it receives as soon as practicable and in any event within two (2) Business Days of receiving a request for information;
- 11.6.2 provides the Client with a copy of all Confidential Information in its possession, or power in the form that the Client requires within five (5) Business Days (or such other period as the Client may specify) of the Client's request; and

- 11.6.3 provides all necessary assistance as reasonably requested by the Client to enable the Client to respond to the request for information within the time for compliance set out in section 10 of the Freedom of Information Act 2000 ("FOIA") or regulation 5 of the Environmental Information Regulations ("EIR").
- 11.7 The Contractor does not respond directly to a request for information unless authorised to do so by the Client.
- 11.8 The Client may disclose the Confidential Information:
 - on a confidential basis to any central government body for any proper purpose of the Client or of the relevant central government body, which shall include: disclosure to the Cabinet Office and/or HM Treasury for the purpose of ensuring effective cross-Government procurement processes, including value for money and related purposes;
 - 11.8.2 to Parliament and Parliamentary Committees or if required by any Parliamentary reporting requirement;
 - 11.8.3 to the extent that the Client (acting reasonably) deems disclosure necessary or appropriate in the course of carrying out its public functions;
 - 11.8.4 on a confidential basis to a professional adviser;
 - on a confidential basis for the purpose of the exercise of its rights under the Contract; or
 - on a confidential basis to a proposed body in connection with any assignment, novation or disposal of any of its rights, obligations or liabilities under the Contract,
 - and for the purposes of the foregoing, references to disclosure on a confidential basis shall mean disclosure subject to a confidentiality agreement or arrangement containing terms no less stringent than those placed on the Client under this provision.
- 11.9 Before sharing any Confidential Information in accordance with Clause 11.8 above, the Client may redact the Confidential Information. Any decision to redact information made by the Client shall be final.
- 11.10 The Contractor ensures that all Confidential Information is retained for disclosure throughout the period for retention and permits the Client to inspect such records as and when reasonably requested from time to time.
- 11.11 The Client shall not be in breach of the Contract where it can show that any disclosure of Confidential Information is made solely and to the extent necessary to comply with the FOIA or the EIR. To the extent permitted by the time for compliance under FOIA or the EIR 2004, the Client shall consult the Contractor where the Client is considering the disclosure of Confidential Information under FOIA or the EIR 2004 and, in any event, shall provide prior notification to the Contractor of any decision to disclose the Confidential Information. The Contractor acknowledges and accepts that its representations on disclosure during consultation may not be determinative and that the decision whether to disclose Confidential Information in order to comply with FOIA

or the EIR 2004 is a matter in which the Client shall exercise its own discretion, subject always to the provisions of FOIA or the EIR 2004. For the avoidance of doubt, nothing in this Clause shall affect the Contractor's rights at law.

11.12 Nothing in this Clause shall affect the parties' obligations of confidentiality where information is disclosed orally in confidence.

12 <u>Data protection</u>

12.1 In this clause 12 and where used elsewhere in this Contract, the following terms shall have the following meanings:

"Contractor Staff" all directors, officers, employees, agents, Contractors and contractors of the Contractor and/or any Sub-Processor engaged in the performance of its obligations under this Contract;

"Controller" the same meaning as given in Data Protection Legislation; "Data Loss Event" any event that results, or may result, in unauthorised access to Personal Data held by the Contractor under this Contract, and/or actual or potential loss and/or destruction of Personal Data in breach of this Contract, including any Personal Data Breach; "Data Protection Legislation"

- (a) the Data Protection Act 2018;
- (b) Regulations made under the Data Protection Act 2018;
- (c) the UK GDPR; and
- (d) all applicable Laws and regulations relating to Processing of Personal Data, including where applicable the guidance and codes of practice issued by the Information Commissioner or relevant Government department in relation to such legislation, in

each case as amended from time to time;

"Data Subject" the meaning given to it in the Data Protection Legislation, as amended from time to time:

"Data Subject Access Request" a request made by a Data Subject in accordance with rights granted pursuant to the Data Protection Legislation to access their Personal Data:

"Personal Data" any information relating to an identified or identifiable natural person; an identifiable natural person is one who can be identified, directly or indirectly, in particular by reference to an identifier such as a name, an identification number, location data, an online identifier or to one or more factors specific to the physical, physiological, genetic, mental, economic, cultural or social identity of that natural person;

"Personal Data Breach" the meaning as given in the Data Protection Legislation;

"**Processing**" the meaning given to it in the Data Protection Legislation but, for the purposes of this Contract, it shall include both manual and automatic processing and "**Processed**" shall be interpreted accordingly;

"**Processor**" the meaning given to it in the Data Protection Legislation, as amended from time to time:

"Protective Measures" appropriate technical and organisational measures which shall be sufficient to secure that the Processor will meet the requirements of the Data Protection Legislation and ensure the protection of the rights of the Data Subject and may include (without limitation):

- (a) Pseudonymisation and encrypting Personal Data;
- (b) ensuring on-going confidentiality, integrity, availability and resilience of systems and services used for data processing;
- (c) measures to restore the availability and access to Personal Data in a timely manner in the event of a physical or technical incident;
- (d) ensuring that availability of and access to Personal Data can be restored in a timely manner after an incident; and
- (e) regularly assessing and evaluating the effectiveness of such measures adopted by it;

"Pseudonymisation" the same meaning as given in Data Protection Legislation;

"Restricted Country" any country which:

- (a) is outside the United Kingdom;
- (b) is not covered by UK adequacy regulations issued under section 17A of the Data Protection Act 2018 or paragraphs 4 and 5 of Schedule 21 of the Data Protection Act 2018; and
- (c) has not been confirmed by the Client as a non-Restricted Country in writing from time-to-time:

"Special Categories of Personal Data" the meaning given in the Data Protection Legislation;

"Sub-Processor" any third party appointed to Process Personal Data on behalf of the Contractor related to this Contract; and

"UK GDPR" the retained European Union law version of the General Data Protection Regulation (EU) 2016/679 as modified by Schedule 1 to the Data Protection, Privacy and Electronic Communications (Amendments etc.) (EU Exit) Regulations 2019 as in force in the UK from time to time.

12.2 Where any Personal Data is Processed in connection with the exercise of the Parties' rights and obligations under this Contract, the Parties acknowledge that the Client is the Controller and that the Contractor is the Processor.

12.3 The Contractor shall:

- ensure that it and the Contractor Staff comply with all of the applicable requirements of the Data Protection Legislation and shall not knowingly or negligently by any act of omission, place the Client in breach or potential breach of Data Protection Legislation;
- 12.3.2 process the Personal Data only to the extent and in such manner as is necessary for the purposes specified in this Contract and in accordance with instructions from the Client to perform its obligations under this Contract in accordance with Schedule 6 (Personal Data and Data Subjects), unless otherwise required by Law. In such case, the Contractor shall inform the Client of that legal requirement unless the Law prevents such disclosure on the grounds of public interest;
- 12.3.3 notify the Client immediately if it considers that any of the Client's instructions infringe the Data Protection Legislation;
- maintain complete and accurate records and information to demonstrate its compliance with its Processor obligations under this clause 12;
- 12.3.5 keep a record of any Processing of Personal Data it carries out on behalf of the Client including the records specified in Article 30(2) of the UK GDPR and upon request provide a copy to the Client;
- ensure that at all times it has in place appropriate Protective Measures to guard against a Data Loss Event, which the Client may reasonably reject (but failure to reject shall not amount to approval by the Client of the adequacy of the Protective Measures) and which shall be implemented at the Contractor's own expense and at no cost to the Client;
- 12.3.7 not disclose or transfer the Personal Data to any third party or Contractor Staff unless necessary for the provision of the service and, for any disclosure or transfer of Personal Data to any third party, obtain the prior written consent of the Client (save where such disclosure or transfer is specifically authorised under this Contract);
- 12.3.8 take reasonable steps to ensure the reliability and integrity of any Contractor Staff who have access to the Personal Data and ensure that the Contractor Staff:
 - (i) are aware of and comply with the Contractor's duties under this Contract;
 - (ii) are informed of the confidential nature of the Personal Data and do not publish, disclose or divulge any of the Personal Data to any third party unless directed in writing to do so by the Client or as otherwise permitted by this Contract; and
 - (iii) have undergone adequate training in the use, care, protection and handling of Personal Data;

- 12.3.9 notify the Client immediately if it becomes aware of a Data Loss Event or if it receives:
 - (i) from a Data Subject (or third party on their behalf) a Data Subject Access Request (or purported Data Subject Access Request) a request to rectify, block or erase any Personal Data or any other request, complaint or communication relating to the Client's obligations under the Data Protection Legislation;
 - (ii) any communication from the Information Commissioner or any other regulatory authority in connection with Personal Data; or
 - (iii) a request from any third party for disclosure of Personal Data where compliance with such request is required or purported to be required by Law:
- 12.3.10 provide the Client with full co-operation and assistance (within the timescales reasonably required by them) in relation to any complaint, communication or request made (as referred to at Clause 12.3.9) at no cost to the *Client* including by promptly providing:
 - (i) the Client with full details and copies of the complaint, communication or request;
 - (ii) where applicable, such assistance as is reasonably requested by the Client to enable them to comply with the Data Subject Access Request within the relevant timescales set out in the Data Protection Legislation; and
 - (iii) the Client, on request, with any Personal Data it holds in relation to a Data Subject;
- 12.3.11 if requested by the Client, provide a written description of the measures that the Contractor has taken and the Protective Measures in place, for the purpose of compliance with its obligations pursuant to this Clause 12.3 and provide to the Client copies of all documentation relevant to such compliance including, protocols, procedures, guidance, training and manuals; and
- 12.3.12 at the written direction of the Client, delete or return Personal Data (and any copies of it) using a secure method of transfer to the Client on expiry or earlier termination of this Contract unless the Contractor is required by Law to retain Personal Data.
- 12.4 The Contractor shall not Process or otherwise transfer any Personal Data in or to a Restricted Country. If, after the Commencement Date, the Contractor or any Subcontractor wishes to Process and/or transfer any Personal Data in or to any Restricted Country, the following provisions shall apply:
 - 12.4.1 the Contractor shall give a written request in accordance with Clause 12.4.1 ("Third Country Processing Request") to the Client which the Client shall consider in its absolute discretion:

- the Contractor shall set out in its proposal to the Client for a Third Country Processing Request details of the following:
 - the Personal Data which will be transferred to and/or Processed in or to any Restricted Countries;
 - (ii) the Restricted Countries to which the Personal Data will be transferred and/or Processed;
 - (iii) any Subcontractor or other third parties who will be Processing and/or receiving Personal Data in Restricted Countries; and
 - (iv) how the Contractor will ensure an adequate level of protection and adequate safeguards in respect of the Personal Data that will be Processed in and/or transferred to Restricted Countries (in accordance with Article 46 of the UK GDPR and/or Data Protection Legislation) so as to ensure the Client's compliance with the Data Protection Legislation;
- in providing and evaluating the Third Country Processing Request, the Parties shall ensure that they have regard to and comply with the Data Protection Legislation and then-current Client, Contracting Authority and Information Commissioner Office policies, procedures, guidance and codes of practice on, and any approvals processes in connection with, the Processing in and/or transfers of Personal Data to any Restricted Countries; and
- the Contractor shall comply with such other instructions and shall carry out such other actions as the Client may notify in writing, including:
 - (i) incorporating standard and/or model clauses (which are approved as offering adequate safeguards under the Data Protection Legislation) into this Contract or a separate data processing agreement between the Parties; and
 - (ii) procuring that any Subcontractor or other third party who will be Processing and/or receiving or accessing the Personal Data in any Restricted Countries either enters into:
 - (A) a direct data processing agreement with the Client on such terms as may be required by the Client; or
 - (B) a data processing agreement with the Contractor on terms which are equivalent to those agreed between the Client and the Subcontractor relating to the relevant Personal Data transfer, and

in each case which the Contractor acknowledges may include the incorporation of model contract provisions (which are approved as offering adequate safeguards under the Data Protection Legislation) and technical and organisation measures which the *Client* deems necessary for the purpose of protecting Personal Data.

- 12.5 The Contractor shall use reasonable endeavours to assist the Client to comply with any obligations under the Data Protection Legislation and shall not perform its obligations under this Contract in such a way as to cause the Client to breach any of their obligations under the Data Protection Legislation to the extent the Contractor is aware, or ought reasonably to have been aware, that the same would be a breach of such obligations.
- 12.6 The Contractor shall allow for audits of its Processing activity by the Client or the Client's designated auditor and comply with all reasonable requests or directions by the Client to enable the Client to verify that the Contractor is in full compliance with its obligations under this Contract.
- 12.7 The Contractor shall designate a data protection officer if required by the Data Protection Legislation.
- 12.8 Before allowing any Sub-Processor to process any Personal Data related to this Contract, the Contractor shall:
 - 12.8.1 notify the Client in writing of the intended Sub-Processor and processing;
 - 12.8.2 obtain the written consent of the Client; and
 - 12.8.3 enter into a written agreement with the Sub-Processor which gives effect to the terms set out in this clause 12 such that they apply to the SubProcessor; and provide the Client with such information regarding the SubProcessor as they may reasonably require.
- 12.9 The Contractor shall remain fully liable for all acts or omissions of any Sub-Processor.
- 12.10 The Contractor indemnifies the Client against all costs, charges, expenses (including legal and administrative expenses) and payments made by the Client arising out of or in connection with any claim or proceedings under any Data Protection Legislation resulting from any act or omission of the Contractor or any Sub-Processor.
- 12.11 Within thirty (30) Working Days of the date of termination or expiry of this Contract, the Contractor shall return to the *Client* any Personal Data that are Processed in connection with the exercise of the Parties' rights and obligations under this Contract (unless applicable Law requires storage of the Personal Data) and data belonging to the *Client* that is in the Contractor's possession, power or control, either in its then current format or in a format nominated by *Client*, save that it may keep one (1) copy of any such data or information for a period of up to twelve (12) months to comply with its obligations under this Contract, or such period as is necessary for such compliance.

13 Modern slavery, child labour and inhumane treatment

- 13.1 In performing its obligations under this Contract, the Contractor shall at all times:
 - 13.1.1 comply with all applicable anti-slavery and human trafficking laws and regulations from time to time in force including the Modern Slavery Act 2015 and all related guidance issued by the UK government;
 - 13.1.2 comply with the ethical labour policy of the Client or, alternatively, have and maintain its own policies and procedures which achieve, as a minimum, those standards described in the ethical labour policy;

- 13.1.3 comply with the modern slavery compliance obligations notified to the Contractor by the Client; and
- 13.1.4 include in its contracts with all of its subcontractors and suppliers obligations equivalent to those set out in this Clause 13.
- 13.2 The Contractor shall implement full and appropriate due diligence processes and procedures for its Subcontractors, suppliers and other participants in its supply chains so as to ensure that there is no slavery, human trafficking or otherwise forced or coerced labour in its supply chains.
- 13.3 The Contractor shall deliver to the Client on or around each anniversary of this Contract:
 - 13.3.1 written confirmation of its continued compliance with all of the requirements of this Clause 13: and
 - 13.3.2 if requested by the Client, a report setting out the steps it has taken to ensure that modern slavery and human trafficking is not taking place in any of its supply chains or in any part of its business.
- 13.4 The Contractor shall notify the Client immediately it becomes aware of any actual or suspected slavery, human trafficking, forced or coerced labour in a supply chain that has a connection with this Contract or any breach, or potential breach, of any of the requirements of this Clause 13.
- 13.5 If the Client identifies or becomes aware of any potential or actual breach of this Clause 13, the Client shall raise this with the Contractor and the Contractor will immediately implement a remedial plan to ensure that there are no further breaches of this Clause 13. Notwithstanding this, it is acknowledged and agreed that the Client may, at its option and in its absolute discretion, terminate this Contract with immediate effect by giving written notice to the Contractor.

14 Corrupt gifts and payments of commission

- 14.1 The Contractor shall not do, and warrants that in entering this Contract it has not done any of the following (hereafter referred to as 'prohibited acts'):
 - 14.1.1 offer, promise or give to any Crown servant any gift or financial or other advantage of any kind as an inducement or reward;
 - 14.1.2 doing or not doing (or for having done or not having done) any act in relation to the obtaining or execution of this or any other contract with the Crown;
 - 14.1.3 for showing or not showing favour or disfavour to any person in relation to this or any other contract with the Crown; or
 - 14.1.4 enter into this or any other contract with the Crown in connection with which commission has been paid or has been agreed to be paid by it or on its behalf, or to its knowledge, unless before this Contract is made particulars of any such commission and of the terms and conditions of any such agreement for the payment thereof have been disclosed in writing to the Client.

- 14.2 If the Contractor, its employees, agents or any subcontractor (or anyone acting on its behalf or any of its or their employees) does any of the prohibited acts or commits any offence under the Bribery Act 2010 with or without the knowledge or authority of the Contractor in relation to this Contract or any other contract with the Crown, the Client shall be entitled:
 - 14.2.1 to terminate this Contract and recover from the Contractor the amount of any loss resulting from the termination;
 - 14.2.2 to recover from the Contractor the amount or value of any such gift, consideration or commission; and
 - 14.2.3 to recover from the Contractor any other loss sustained in consequence of any breach of this Clause, where this Contract has not been terminated.
 - 14.2.4 In exercising its rights or remedies under this Clause, the Client shall:
 - 14.2.5 act in a reasonable and proportionate manner having regard to such matters as the gravity of, and the identity of the person performing, the prohibited act or committing of any offence under the Bribery Act 2010;
 - 14.2.6 give all due consideration, where appropriate, to action other than termination of this Contract, including (without being limited to):
 - (i) requiring the Contractor to procure the termination of a subcontract where the prohibited act or committing of any offence under the Bribery Act 2010 is that of a Subcontractor or anyone acting on its or their behalf;
 - (ii) requiring the Contractor to procure the dismissal of an employee (whether its own or that of a Subcontractor or anyone acting on its behalf) where the prohibited act or committing of any offence under the Bribery Act 2010 is that of such employee.
- 14.3 Recovery action taken against any person in Her Majesty's service shall be without prejudice to any recovery action taken against the Contractor pursuant to this Clause.

15 Conflicts of interest

- 15.1 The Contractor shall take appropriate steps to ensure that the Contractor is not placed in a position where, in the reasonable opinion of the Client:
 - 15.1.1 there is or may be an actual conflict or potential conflict, between the pecuniary or personal interests of the Contractor and the duties owed to the Client under the provisions of this Contract or any Task Order; or
 - the behaviour of the Contractor is not in the Client's best interests or might adversely affect the Client's reputation.
- 15.2 The Contractor shall as soon as reasonably practicable disclose to the Client full particulars of any behaviour or relationship which might give rise to an actual or potential conflict.

15.3 The Client may terminate this Contract by notice in writing and/or take such other steps it deems necessary where, in the reasonable opinion of the Client, there is or may be an actual conflict or potential conflict, between the financial or personal interests of the Contractor and the duties owed to the Client under the provisions of this Contract. The actions of the Client pursuant to this Clause do not prejudice or affect any right of action or remedy which has accrued or will accrue to the Client.

16 <u>Transparency</u>

- 16.1 Notwithstanding any other term of this Contract (including Clause 11) where applicable and subject to 16.2 and 16.2, the Contractor understands that the Client may publish the Transparency Information to the general public. The Contractor shall assist and co-operate with the Client to enable the Client to publish the Transparency Information.
- 16.2 Before publishing the Transparency Information to the general public in accordance with Clause 16.1 above the Client shall redact any information (which may include the Contractor's Confidential Information) that would be exempt from disclosure if it was the subject of a request for information under the FOIA or the EIR.
- 16.3 The Client may consult with the Contractor before redacting any information from the Transparency Information in accordance with Clause 16.2 above. The Contractor acknowledges and accepts that its representations on redactions during consultation may not be determinative and that the decision whether to redact information is a
 - matter in which the Client shall exercise its own discretion, subject always to the provisions of the FOIA or the EIR.
- 16.4 For the avoidance of doubt, nothing in this Clause 16 shall affect the Contractor's rights at law.

17 Publicity

17.1 The Contractor may only publicise the Works and/or this Contract with the Client's written agreement.

18 Retention and supply of records and data

- 18.1 The Contractor shall maintain all records specified in and connected with this Contract (expressly or otherwise) and make them available to the Client when requested on reasonable notice.
- 18.2 The Contractor shall also permit access to relevant records that relate to the contractual obligations to provide Works under this Contract, held by or controlled by them and reasonably required by the Comptroller and Auditor General, its staff and any appointed representative of the National Audit Office, and provide such explanations and information as reasonably necessary for the following purposes to enable the National Audit Office to carry out:
 - 18.2.1 the Client's statutory audits and to examine and/or certify the Client's annual and interim report and accounts; and
 - 18.2.2 an examination pursuant to Part II of the National Audit Act 1983 of the economy, efficiency and effectiveness with which the Client has used its resources.

- 18.3 With regard to the records made available to the Client under Clause 18.1 of this Clause, and subject to the provisions of Clause 11 of this Contract (Disclosure of Information), the Contractor shall permit records to be examined and if necessary copied, by the Client, as the Client may require.
- 18.4 Unless this Contract specifies otherwise the records referred to in this Clause shall be retained for a period of at least 6 years from the:
 - 18.4.1 end of this Contract term;
 - 18.4.2 termination of this

Contract; or 18.4.3 final payment,

whichever occurs latest.

19 Change of control of the Contractor

- 19.1 The Contractor shall notify the Client, as soon as practicable, in writing of any intended, planned or actual change in control of the Contractor. The Contractor shall not be required to submit any notice which is unlawful or is in breach of either any pre-existing non-disclosure agreement or any regulations governing the conduct of the Contractor in the UK or other jurisdictions where the Contractor may be subject to legal sanction arising from issuing such a notice.
- 19.2 For the purposes of this Clause "**control**" means the power of a person to secure that the affairs of the Contractor are conducted in accordance with the wishes of that person by:
 - 19.2.1 means of the holding of shares, or the possession of voting powers in, or in relation to, the Contractor; or
 - 19.2.2 virtue of any powers conferred by the constitutional or corporate documents, or any other document, regulating the Contractor,

and a change of control occurs if a person who controls the Contractor ceases to do so or if another person acquires control of the Contractor.

19.3 The Client shall consider the notice of change of control and advise the Contractor in writing of any concerns the Client may have.

20 **Equality**

- 20.1 The Contractor shall not unlawfully discriminate either directly or indirectly on the grounds of age, disability, gender (including re- assignment), sex or sexual orientation, marital status (including civil partnerships), pregnancy and maternity, race, or religion or belief.
- 20.2 Without prejudice to the generality of the obligation in Clause 20.1 above, the Contractor shall not unlawfully discriminate within the meaning and scope of the Equality Act 2010 (or any statutory modification or re- enactment thereof) or other relevant or equivalent legislation in the country where the contract is being performed.

- 20.3 The Contractor agrees to take reasonable efforts to secure the observance of the provisions of this Clause by any of its employees, agents or other persons acting under its direction or control who are engaged in the performance of this Contract.
- 20.4 The Contractor agrees to take reasonable efforts to reflect this Clause in any subcontract that it enters into to satisfy the requirements of this Contract and to require its subcontractors to reflect this Clause in their subcontracts that they enter into to satisfy the requirements of this Contract.

21 <u>Dispute resolution</u>

21.1 The provisions of Secondary Option W2 shall apply to any dispute resolution proceedings under this Contract.

22 Notices

- 22.1 Any notices sent under this Contract must be in writing and in accordance with this clause 22.
- 22.2 The following table sets out the method by which notices may be served under this Contract and the respective deemed time and proof of service:

Manner of Delivery	Deemed time of service	Proof of service		
Email	9.00am on the first Business Day after sending.	Dispatched as a pdf attachment to an e-mail to the correct e-mail address without any error message		
Personal delivery	On delivery, provided delivery is between 9.00am and 5.00pm on a Business Day. Otherwise, delivery will occur at 9.00am on the next Business Day.	Properly addressed and delivered as evidenced by signature of a delivery receipt.		
Prepaid, Royal Mail Signed For™ 1st Class or other prepaid, next working day service providing proof of delivery	At the time recorded by the	Properly addressed prepaid and delivered as evidenced by signature of a delivery receipt.		

22.3 Notices shall be sent to the addresses set out below or at such other address as the relevant Party may give notice to the other Party for the purpose of service of notices under this Contract:

Contracto	r	Client
0011114010	•	O II O I I

Contact	VINCI Construction REDACTED	REDACTED
Address	Astral House, Imperial Way Watford Hertfordshire WD24 4WW	DWP, Caxton House Tothill Street London SW1H 9NA
Email	REDACTED	REDACTED

23 Parent Company Guarantee and Performance Bond

- 23.1 The Contractor shall provide the Parent Company Guarantee, duly executed by the Parent Company, within fifteen (15) Business Days of any request made the Client.
- 23.2 If so requested by the Client in relation to any Task Order, the Contractor shall provide the Performance Bond, duly executed by the bondsman, within fifteen (15) Business Days of any such request made the Client.

24 <u>Client's Schedules</u>

24.1 To the extent relevant to this Contract and any Task Order, the Contractor shall comply with the Client's Schedules.

25 Miscellaneous

- 25.1 This Contract and each Task Order embody the complete and entire agreement between the Client and the Contractor in relation to the Project and supersede all other oral and/or written communications. The parties shall not be bound by, or be liable for,
 - any statement, representation, promise, inducement or understanding not set forth herein.
- 25.2 Save for any Task Orders issued in accordance with this Contract, no amendments of or modifications to this Contract shall be valid unless reduced to writing and executed as a deed by both parties.
- 25.3 Save where expressly stated, the Contracts (Rights of Third Parties) Act 1999 is excluded from applying to this Contract and nothing herein, save where expressly stated, confers or purports to confer to any third party any benefit or any right to enforce any term and/or condition of this Contract.
- 25.4 This Contract shall be governed by and construed and interpreted in accordance with English law and be subject to the jurisdiction of the courts of England and Wales.
- 25.5 This Contract is executed as a deed and was delivered when it was dated.

```
The corporate seal of Secretary of State for )
Work and Pensions is hereunto ) affixed and authenticated by: )
)
```

SEAL OF SECRETARY OF STATE FOR WORK AND PENSIONS

Executed as a deed by VINCI Construction UK Ltd acting by either two directors or one director and the company secretary:

REDACTED REDACTED

Director's name (CAPITAL LETTERS)

Director's signature

REDACTED

Director's/Company secretary's name Director's/Company secretary's signature

(CAPITAL LETTERS)

Appendix 1

Task Orders

The following shall be incorporated as part of the conditions of contract and in the event of a conflict with any other term, the terms of this Appendix shall take precedence:

The purpose of this Appendix 1 is to confirm when and how the Contractor will prepare a Task Order Proposal and how the approvals process for a Proposed Task will operate.

Part A

1 Definitions

- 1.1 In paragraphs 1 to 7 the following definitions shall apply:
- (a) "Allocation Order" means the order in which the Delivery Contractors are ranked as a result of the Procurement Process, as set out in Part C of this Appendix 1 and as may be amended from time to time in accordance with the provisions of Part A of this Appendix 1;
- (b) "Delivery Contractors" means the delivery contractors identified in Appendix 7 that the Client has entered into a Lot Contract with to deliver the construction works and

a s

s

0

С

i

а

t

е

d

s e

r

i

С

е

S

i

n

Ε

n

g

ı

а

n

d

p

r

t

Α

R

u

n

0

f

f

D

е

ı

i

٧

е

r

У

C

0

n

t

r

а

С

t

0

r

m

е

а

n

а

D

е

I

i

٧

е

r

У

С

0

n

t

r

а

С

- or and any other contractors more particularly specified by the Client and any Replacement Delivery Contractor (each a "**Delivery Contractor**");
- (c) "Extraordinary Event" means where the Client exercises its rights under clause 5.2.4 of any Lot Contract in relation to a Notifiable Event (as defined therein) in relation to any relevant Task Order:
- (d) "Extraordinary Identification Procedure" means the process of selecting a Delivery Contractor to participate in the Pricing Process where an Extraordinary Event arises in accordance with paragraph 4 of this Part A; and
- (e) "Identification Procedure" means either the Rotational Procedure, the

Extraordinary Identification Procedure or the Run Off Identification Procedure:

- (f) "Ineligible Delivery Contractor" means for each Task Order where the Rotational Procedure applies, a Delivery Contractor:
 - (i) that has not selected the relevant NUTS Code(s) for that Task Order and/or the Client has not approved the selection of a NUTS Code pursuant to paragraph 1.28 of Appendix 4; or
 - (ii) where a Suspension Event applies to a Delivery Contractor at the time such Task Order is placed and the Client, in its sole discretion, chooses not to waive such Suspension Event;
- (g) "Lot Contract" means any lot contract entered into by the Client with a Delivery Contractor pursuant to the Procurement Process;
- (h) "Pricing Process" means the pricing of a Task Order to a Delivery Contractor in accordance with the provisions of Part B of this Appendix 1;
- (i) "Procurement Process" means the procurement process relating to the CWAS Framework pursuant to which the Delivery Contractors are selected to enter into a Lot Contract;
- (j) "Replacement Delivery Contractor" means a replacement Delivery Contractor appointed in accordance with paragraph 3.6 of Part A of this Appendix 1;
- (k) "Rotational Procedure" means the procedure of identifying a Delivery Contractor to participate in the

3

- or that previously provided works or services to the Client at the relevant Site:
- (m) "Run Off Identification Procedure" means the process of selecting a Delivery Contractor to participate in the Pricing Process where a Task Order relates to Run Off Works, in accordance with paragraph 3 of this Part A;
- (n) "Run Off Longstop Date" means the date 3 months after the date of this Contract;
- (o) "Run Off Period" means the period of 3 months immediately preceding the date of this Contract;
- (p) "Run Off Task Order" means any Task Order for any Run Off Works:
- (q) "Run Off Works" means any works or services to be instructed under any Task Order which directly relate to any works or services that a Delivery Contractor previously provided to the Client at a Site during the Run Off Period;
- (r) "Suspension Event" means where:
 - (i) a Critical Failure Event occurs; and/or
 - (ii) either the Client or the Delivery Contractor refers a dispute under this Contract and/or any Task Order to the Adjudicator in accordance with clause W2.2 of Secondary Option W2; and
- (s) "Worked Example" means the worked example set out at in Part D of this Appendix 1.

2 <u>Identification Procedure</u>

2.1 An Identification Procedure shall be used to select a Delivery Contractor to participate in the Pricing Process.

3 Rotational Procedure

- 3.1 On the commencement of the Delivery Term, the Allocation Order shall apply.
- 3.2 Subject to paragraph 3.7 of this Appendix 1, where the Client issues a Works

Task Order provided that the Works Task Order relates to the Works set

that Services Task Order, the Rotational Procedure applies to the allocation of Task Orders.

	Contractor to participate in the Pricing Process for that Task Order.	T h	f O
3.4 Su	bject to paragraph 3.4 of Part A of this Appendix 1, where Rotational Procedure applies:	e R o	I I 0
Client shall select	3.4.1 for the first Task Order issued under this Lot, the	t a t	w i n
	the Delivery Contractor listed first on the Allocation Order to participate in the Pricing Process;	i o n	g t h
Lot, the Client	3.4.2 for each subsequent Task Order issued under this	a I P	e a ı
	shall select the next applicable listed Delivery Contractor to participate in the Pricing Process; and	r O C	0 C
	3.4.3 the Client shall not be required to select an Ineligible Delivery Contractor,	e d	a t
	as illustrated in the Worked Example.	u r e	o n
3.5 A	Delivery Contractor may be excluded by the Client from participating in the Rotational Procedure:	s h a	o f a
	3.5.1 for the next applicable Task Order following receipt of a YELLOW Score; and	l l n	S e r
	3.5.2 all applicable Task Orders:	o t	v i
(i)	following receipt of a RED Score until such time as the Delivery Contractor demonstrates to the Client's reasonable satisfaction that the Delivery Contractor has	a p p I	c e s T
	successfully addressed the underlying cause(s) of the RED Score; and	y t o	a s k
(ii)	following the occurrence of a Suspension Event for such time as the Suspension Event persists.	a T a	O r d
3.6	If the Client terminates a Lot Contract, the Client may remove that Delivery Contractor from the Allocation Order and it shall be entitled to enter into a Lot	s k O r d	e r t o
	Contract with a replacement contractor (a "Replacement Delivery Contractor"). Where a Replacement Delivery Contractor is appointed, then that Replacement Delivery Contractor shall be added to the bottom of the Allocation		a D e I i
	Order and the Client shall issue a revised Allocation Order to each Delivery Contractor and such revised Allocation Order shall supersede any previous version for the purposes of the Rotational Procedure.	s u e d	v e r y

5

2

F

0

r

а

n

У

е

Х t

r

а

0

r d

n

а

r

У

е

٧

е

n

t

а

n d

0

r t

а

s k

0 r

d е

r

h i

С

h 0

С

С

u

r s u

n

d

е

r

а

n

0

t

h

е

L

0

C

0

n

t

r

а

С

t

W

h

е

r

е

5.2

Contractor or	any Task C	rde	r to	which	the	Extraordinary
Identification	Procedure	or	the	Run	Off	Identification
Procedure applies.						

Extraordinary Identification Procedure and Run Off Identification Procedure

- 4.1 Where an Extraordinary Event arises the Client shall, at its discretion, be entitled to identify the Delivery Contractor it considers to be the most suitable for the procurement irrespective of the Allocation Order.
- 4.2 Where a Task Order relates to Run Off Works, the Client may, at its discretion, be entitled to allocate the Run Off Works to the appropriate Run Off Delivery Contractor provided that:
 - 4.2.1 the Client can demonstrate that this may be of benefit to the Client (including but not limited to enable the Client to receive the benefit of cost savings); and
 - 4.2.2 the Identification Procedure for the relevant Task Order takes place on or before the Run Off Longstop Date.

5 Alternative Lot Contracts

- 5.1 For any Extraordinary Event and/or Task Order where:
 - 5.1.1 no Delivery Contractor wishes to be selected for the Pricing Process or to undertake the Works that are the subject of the Extraordinary Event: and/or
 - 5.1.2 the Client determines that each applicable Delivery

Contractor:

is an Ineligible Delivery Contractor; (i) and/or (ii) has received YELLOW score and/or a RED score,

so that they are excluded from participating in the Rotational Procedure for that Task Order,

the parties agree that the Client, at its discretion, shall be entitled to select a delivery contractor from an alternative Lot Contract and the Client shall apply the applicable identification procedure set out within that Lot Contract.

S

s

u

е

d

р

u

r

s

u

а

n t

t

0

р

а

r

а

g

r

а

р

h

1

t

h

е

C

ı

i

е

n

t

s

h

а

ı

ı

b

е

е

n

t

i

t

Ι

е

d

t

0

1

F

O

Ι

0

w

i

n

g

r

е

С

е

i

р

t

0

f

а

Т

а

s

k

0

r

d

е

r

Ρ

r

0

р

0

s

а

ı

i

- er that other Lot wishes to be selected to undertake the works that are the subject of the extraordinary event; and/or
- the Client determines that each applicable Order

 delivery contractor appointed under a Lot Propos
 Contract in relation to that other Lot: (i) is an ineligible delivery contractor; and/or

 propos

(ii) has received a YELLOW score and/or a RED score,

Approv
al of
Task
Order
Propos
als and
the
Propos
ed
Task

2

so that they are excluded from participating in the rotational procedure under that other Lot Contract for that Task Order.

the parties agree that the Client, at its discretion, shall be entitled to use the applicable identification procedure under this Lot to select a Delivery Contractor to deliver the applicable works.

6 Pricing Process

6.1 Following the completion of the relevant Identification Procedure which results in a Delivery Contractor being eligible to participate in the Pricing Process, the Client shall undertake the Pricing Process with that Delivery Contractor.

Part B

1 Preparation of Task Order Proposal

- 1.1 The Client shall submit a written request (the "Project Instruction") for the Delivery Contractor to produce outline proposals for the Proposed Task. The Client shall provide such information (including arrangements to access the Site) as may be reasonably required by the Delivery Contractor to enable it to develop a Task Order Proposal, provided that the Client shall not be obliged to provide such information where this may cause the Client to be in breach of Law and/or where such information is otherwise confidential to the Client.
- 1.2 Within ten (10) Business Days (or such other period as is agreed by the Parties) of receipt by the Delivery Contractor of a Project Instruction, the Delivery Contractor shall submit a Task Order Proposal in respect of the Proposed Task together with such additional information/documentation as is required.

2

equest any reasonable additional information and/or any reasonable assistance (as the case may be) from the Delivery Contractor which the Client considers appropriate to enable it to decide whether to approve the relevant Task Order Proposal and the Delivery Contractor shall provide such information and/or any assistance (as the case may be) within three (3) Business Days of receipt of the relevant request for such further information and/or and assistance (as the case may be). The Delivery Contractor acknowledges and agrees that the costs incurred by the Contractor in responding to such requests and/or entering into any communications with the Client and/or incurring any expenditure of whatever nature in relation to such requests shall not entitle the Delivery Contractor to any payment of whatever nature and any such costs shall be borne by the Delivery Contractor unless the Client agrees otherwise in writing.

- 2.2 In deciding whether to approve a Task Order Proposal the Client shall be entitled to take into account all relevant factors which it considers would affect whether the relevant Task Order Proposal is acceptable, including, without limitation:
 - 2.2.1 whether the Task Order Proposal is, in the sole opinion of the Client, affordable and/or offers value for money;
 - 2.2.2 whether the Task Order Proposal, in the sole opinion of the Client, contains sufficient information to enable the Client to approve the relevant Task Order Proposal; and/or
 - 2.2.3 whether the potential impact of the implementation of the Task Order Proposal would, in the sole opinion of the Client, adversely impact on the business and/or activities of the Client (including, without limitation, any of the Client's statutory duties); and/or
 - 2.2.4 whether the implementation of the Task Order

Proposal would

breach any Law, any Client's Policies or any wider government strategy or government policy.

2.3 Within ten (10) Business Days (or such longer period to be agreed by the Parties) of the later of the submission to the Client of a Task Order Proposal and the receipt by the Client of any additional information requested by the Client pursuant to paragraph 2.1, the Client shall notify the Delivery Contractor whether:

tractor shall carry out the Proposed Task as part of a Task Order in accordance with the terms of the approved Task Order Proposal; or

- 2.3.2 the Client rejects the Task Order Proposal and the provisions of paragraph 2.4 shall apply.
- 2.4 The Delivery Contractor acknowledges and agrees that where the Client elects to reject any Task Order Proposal then the Delivery Contractor's costs in relation to the proposal shall be borne by the Delivery Contractor unless the Client agrees otherwise in writing.
- 2.5 The Client may appoint a third party, any other Delivery Contractor and/or any other delivery contractor from an alternative Lot Contract to provide the whole or any part of the Task Order where the Delivery Contractor:
 - 2.5.1 is in material breach of its obligations under this Contract or any Task Order;
- 2.5.2 fails to provide and/or the Client rejects a Task Order Proposal;
 - 2.5.3 confirms that it does not intend to submit a Task Order Proposal or fails to submit a Task Order Proposal following receipt of a Project Instruction within the time period specified in and in accordance with the provisions of this paragraph 2;
 - 2.5.4 the Parties have been unable to agree the whole or any part of a Task Order Proposal; and/or
 - 2.5.5 where the Delivery Contractor has not and/or all of the Delivery Contractors have not selected the relevant NUTS Codes for the whole or any part of a Task Order.
 - 2.6 The Client may appoint any Delivery Contractor from this Lot Contract to provide the whole or any part of a task order which occurs under another Lot Contract where the delivery contractor under that Lot Contract:
 - 2.6.1 is in material breach of its obligations under that Lot Contract or any task order occurring under that Lot Contract;
 - 2.6.2 fails to provide and/or the Client rejects a task order proposal provided under that Lot Contract;
 - 2.6.3 confirms that it does not intend to submit a task order proposal or fails to submit a task order

i

i

s of that Lot Contract;

- 2.6.4 the Parties have been unable to agree the whole or any part of a task order proposal; and/or
- 2.6.5 where, under that Lot Contract, the delivery contractor has not and/or all of the delivery contractors have not selected the relevant NUTS codes for the whole or any part of a task order.
- 2.7 The Delivery Contractor acknowledges and agrees that the Delivery Contractor shall not be entitled to any payment, compensation, damages, costs, losses and/or expenses arising out of or in connection with the circumstances set out in paragraph 2.5.
- 2.8 Where a Task Order Proposal has been approved by the Client (to incorporate any amendments the Client deems necessary), the Client may sign the Task Order Proposal (as amended, if necessary) and issue to the Delivery Contractor and such signed Task Order Proposal shall be deemed to form the Task Order. On receipt of a Task Order pursuant to this paragraph, the works to be carried out and performed by the Delivery Contractor as set out in the Task Order shall be deemed to become part of the Works.
- 2.9 The parties agree that if the Task Order Proposal is approved and any Task Order is issued by the Client and accepted by the Delivery Contractor under this paragraph 2, then the provisions of:

2.9.1 this

Contract; and

2.9.2 the

relevant Task

Order,

shall together govern the carrying out of any Works to be performed by the Delivery Contractor. In the event of any conflict between the documents referred to in this paragraph 2.8, then the order of precedence shall be:

- 2.9.3 the relevant Task Order; and
- 2.9.4 this Contract.
- 2.10 This Contract shall be binding on the parties for the Delivery Term and shall govern any Works provided to the Client

or to the date of this Contract.

- 2.11 At any time, the Client may set off any liability of the Delivery Contractor to the Client against any liability of the Client to the Delivery Contractor, whether either liability:
 - 2.11.1 is liquidated or unliquidated; or
 - 2.11.2 arises under this Contract and/or any Task Order.
- 2.12 Without prejudice to any other rights or remedies which the Client may have, whether under this Contract or under any Task Order, the obligations for the Client and/or the Delivery Contractor to make payments in respect of any Task Orders shall be continuing obligations notwithstanding the termination of this Contract or any individual Task Order.

Part C - Allocation Order - Lot C South

REDACTED	
REDACTED	
REDACTED	
REDACTED	
REDACTED	

Part D - Worked Example

REDACTED

Part E – Amendments to the NEC4 ECC

3 General

3.1 Amend clause 11.1 to add a new sentence at the end: "The Contractor and the

Client agree that the Activity Schedule, Scope, Site Information and Accepted

Programme attached to any Task Order shall be deemed to be the Activity Schedule, Scope, Site Information and Accepted Programme relevant to the works instructed under the relevant Task Order."

- 3.2 Amend clause 11.2(16) to add at the end, on a new line after the third and fourth bullet points, "including the Part A Scope and the Part B Scope."
- 3.3 Add new clause 11.2(21):

0

f

i

t

s

0

b I

i

g

а

t

i

0 6

S

u

n

d

е

r

t

h

е

С

0

n

t

r

а

С

t

а

n

d

0

t

h

е

D

е

i

٧

е

r

У

Α

g

r

e

6

Compe

nsation

<u>events</u>

е

m

е

n

t

and th	ne C	ontract	or d	ated	d th	e c	date	of	this
contra	ct' [<mark>IN</mark>	SERT	DAT	E u	pon	sig	natur	<mark>e</mark>]."	,

- 3.4 Add new clause 11.2(22):
 - "11.2(22) The Part B Scope means the scope of works referred to in the Task Order."
- 3.5 Add new clause 11.2(23):

"11.2(23) Delivery Agreement means the delivery agreement dated [DATE] entered into between the Client and the *Contractor*."

- 3.6 Add new clause 11.2(24):
 - "11.2(24) Task Order means a task order issued by the Client to the Contractor under the Delivery Agreement, such Task Order to comprise the contract.

4 The Contractor's design.

- 4.1 At the end of clause 21.1 delete the full stop and add: "and in the provision of its design and carrying out of due diligence on the Part B Scope acts as a competent and experienced contractor and uses all the reasonable skill, care and diligence and the best up-to-date practice to be expected of a fully qualified and competent designer experienced in designing and performing and carrying out design for projects of a similar size, scope, nature, complexity and timescale to the works."
- 4.2 Add new clause 21.4:

"21.4 The Contractor accepts entire responsibility for the Part A Scope."

5 Requirements for Instructions

- 5.1 Add to the start of 17.1 and 17.2: "Subject to clause 17.3"
- 5.2 Add new clause 17.3: "If the ambiguity, discrepancy, omission, mistake or inconsistency referred to in clause 17.1 is within the Part A Scope, an instruction issued under clause 17.2 relating to such ambiguity, discrepancy, omission, mistake or inconsistency does not give rise to a compensation event and the Contractor is fully responsible for such ambiguity, discrepancy, omission, mistake or inconsistency and there is no effect upon, Defined Cost, Completion and/or meeting a Key Date and the Contractor is not entitled to relief from any

6 Α m е n d С П

Α

1

0 f

t

h

е

ı

n

S

0

ı

е

n

С

y A

С

t

1

9

8

6

R

1

0

В

)

7

а

d

d

а

t

t

h

е

b

е

g

i

n

n

n

g

S

u

b

j

е

С

t

t

0

t

h

е

C

0

r

p o

r

а

t

е

ı

n

s

0

ı

٧

е

n

С

У

а

n

d

G

а

u

s

е

9

1

4

t

0

- 1(1): "an instruction relating to an ambiguity, discrepancy, omission, mistake or inconsistency referred to in clause 17.3."
- 6.2 Add to the end of the seventh bullet point in clause 61.4: "and the Contractor is not entitled to a compensation event to the extent that the Contractor is entitled to a compensation event under a different Task Order for the same event or Lot. The Contract agrees there shall be no double counting in respect of any compensation event entitlement"
- 6.3 Add new bullet point in clause 61.4 after the seventh bullet point: "relates to the carrying out of the Management Services under this Contract between the Client and the Contractor the Contractor shall not be entitled to a compensation event."

7 Termination

- 7.1 Amend clause 90.2 to add at the beginning "Subject to the Corporate Insolvency and Governance Act 2020"
- 7.2 Amend the table at 90.2 reason column for The Contractor. Delete 'R10'. Add 'R10A and R10B'
- 7.3 Amend the table in clause 90.2 reason column for the Client to add in "R23" after "R22".
- 7.4 Amend clause 91 to add a new clause 91.9:
 - "91.9 The Client may terminate if the Client terminates the Delivery Agreement and/or any other Task Order (R23)."
- 7.5 Amend clause 91.1 to add at the beginning "Subject to the Corporate Insolvency and Governance Act 2020",
 - 7.5.1 In the second bullet point:
 - (i) after (R10) delete the full stop and add

",or

provided or taken any step in relation to a Scheme of Arrangement under Part 26 or Part 26A of the Companies Act 2006 but excluding a Scheme of Arrangement as a solvent company for the purposes of amalgamation or re construction (R10A), or

applied to the court for, or obtained, a moratorium under Part

39

overnance Act 2020"

7.7 Amend clause 91.5 to add at the beginning "Subject to the Corporate Insolvency and Governance Act 2020"

S

е

t

0

u

t

i

n

t

h

е

С

0

n

t r

> a c

t

D

а

t

а

С

Ι

а

i

m

е

d

i

n

r

e s

р

е

С

t

0

f

е

а

С

h

C I

а

i

m

0

r

s

е

r

i

е

s

0

f

С

Ι

а

i

m

S

a r

i

S

i

n

g

r

0

m

t

h

е

s

а

m

e

0

r

i

g

n

- 7.8 Amend clause 91.6 to add at the beginning "Subject to the Corporate Insolvency and Governance Act 2020"
- 7.9 Y2.5 (NEC4) To the extent that the Corporate Insolvency and Governance Act 2020 is deemed to apply to the Housing Grants, Construction and Regeneration Act 1996, add at the beginning "Subject to the Corporate Insolvency and Governance Act 2020"

X18 Limitation on Liability

Delete the entirety of clause X18 and insert the following wording:

- "X18.1 Nothing in this contract shall exclude or limit the Contractor's liability for:
 - X18.1 death or personal injury;
 - X18.2 fraud or fraudulent misrepresentation; or
 - X18.3 any negligence of the Contractor or its personnel, agents or subcontractors.
- X18.2 The Contractor's liability to the Client for the Client's indirect or consequential loss is limited to the total of the Prices.
- X18.3 For any one event, the liability of the Contractor to the Client for loss of or damage to the Client's property is limited to £10,000,000.
- X18.4 The Contractor's liability to the Client for Defects due to its design which are not listed or the Defects Certificate is limited to the sum set out in the Contract Data¹ any one claim.
- X18.5 Subject to clause X18.1 and other than the excluded matters identified in clause
 - X18.3, the Contractor's liability under or in connection with this Contract, whether for breach of contract, in tort or for breach of statutory duty, shall be limited to the sum

£5m, Lot B:

£5m and

Lot C:

£10m.

40

¹ 1 Lot A:

	ating or	underlying cause.				
X18.	6 The exclu	ded matters are:		Task C) r đite ne <i>t</i> sio	ostridenci ėsse c
the contract for:	X18.6.1	amounts payable by the	he Contractor as stated in			15 क्रिडोडोडा इस्ट्राइडोडा
	(i)	loss of or damage to and	the Client's property;		of all N	amme") lecessary saofdaorystr
	(ii)	delay damages if Op	tion X7 applies."			rePTearfsdxri00a
		Annex 1		Any ot		vantarnyske
	<u>Tas</u>	sk Order Proposal Temp	<u>olate</u>	Task C		ion free an <i>y dates</i> ar
Project Tit	ele and Add	lress:		met		nal Client
Task Orde	r Proposal	Details	Description	Task C		rting date
Task				Task C	rideoluadi	ictor in coi ne s d et lails :
to be unde	rtaken by th with the Ta dology as to	the Works required the Contractor in the State of the Works will the Works will			limitati minim ct ifrialera such F Order Pi	ed Insura on, the Co um limit of non Aldone Required In roposal D
involved in (such as th	the delivery	or's key personnel or of the Task Order or's project manager, supervisor)	Name: Job:	the wo	or ther 6di or kre sult Order	of any ad சார்!செஞ்ச ி of the imp he sugges
			Responsibilities: Qualifications:	for eac	the exercited and the exercited articles	rp6pme So tent not al er the Con
			Experience:	progra	Project mme fo	tractor is to t Title an acceptan
		or's Sub-Contractors of the Task Order				Order Pro
Order, inclu Contractor	ude details believes wi	Site for the Task of any access that the II be required to ation of the Task		The we	Task (eather meas	re weathe delay dan Drder, calc ata are the urements de where Drder defe

 $^{^{2}}$ Note: if the Task Order is for any design, the description will need to clearly set out what that design

is and what it is for.

3 DWP to confirm rate and if different rates will apply to different Lots. Where appropriate, the rate shall be agreed with reference to a percentage of the Prices.

_	_
Applicable dispute resolution procedure option	NEC4 ECC Option W2 W P
	е Р
Details of the working areas	ac e ee <u>n</u>
	pt <u>d</u>
The key persons are those identified in the	th <u>i</u>
key persons schedule in	<u>-</u>
	e <u>X</u> ter <u>2</u>
The total of the Prices for this Task Order	m
which shall be based upon the Rates and	s <u>d</u>
Prices together with the assumptions behind	of <u>d</u>
these costings	thi n
Details of the cash flow profile for the	s Ta
payment of the Prices.	
paymont of the contest	_ a
	Any compensation events shall be
	assessed in accordance with clause 63.12 t
	of the Terms and Conditions.
	d <u>a</u>
Details of the proposed Task Order fee	d <u>a</u> ag <u>t</u> re <u>a</u>
percentage	
The following documents attached to this Task	
orm part of this Task Order:	to A
•	pr
Activity Schedule	000
·	ee g
Scope	ac ac
Site Information	co h
Accepted Programme • Cash flow prof	5
,	ng 0
	ly. 0
	Executed as e
	CONSTRUCTION
	either two directors
	company secretary
	р
	1
	e
Signed by THE	REDACTED t
SECRETARY OF STATE	Director's name (OA
	d
FOR WORK AND	d
PENSIONS	REDACTED "
REDACTED	Director's/Company
	(CAPITAL LETTERS
Authorised Signatory	3

Task Order process

Name

PART ONE - DATA	A PROVIDED BY THE CLIENT		
			Address Address
1 General			s for for [] commu e
			nicatio
The conditions of co	ontract are the core clauses and the cla	uses for the following	ns e c
	otion for resolving and avoiding disputes		t
of the NEC4 Engine	ering and Construction Contract June 2		r
amendments dated	January 2019.		Addres n
		_	s for
Main Option	A	Option for	
		resolving and	W2
		avoiding disputes	
	ional Completion), X7 (Delay Damages ns (Performance Bond), X14 (X), X16 ([]
Performance Indica		Retention) A20 (Rey	
	Y(UK)2, and any Secondary Options		
	and such Secondary Options shall a only.	ipply to that Task Order	[]
	·y.		
The works are	As set out in the relevant Task Orde	er	
			[]
The Client is	The Secretary of State for Work and	Pensions	
Addres			electro i
s for			nic c
			commu c nicatio o
			ns m
electro nic			m
comm			u —
unicati			The n
ons			Superv i isor is ^C
			a
The Project Manage	er is		t
			<u>,</u> i
			Name [o
			J n s
			Address
			for []The
			commu Scope is
			nc III I OI
			any any

Works instructed under	r a Task Order: as attached Order together with any ot relevant Task Order ⁴				n	conditio to be			(3)
The Site For an relevant Task Informa	y Works instructed under a ition is in Order	Task Ord	der: a	as attached to the		key date		Г	
The boundaries of the relevant Task Ord	For any Works instructed er the site are	under a ⁻	Гask	Order, as set out	in			(1)	For any the releva
The language of the English contract is							3	Time	
							The		_
The <i>law of the</i> contract is the law of	England, subject to the jur	isdiction	of the	e Courts of Englar	nd				
The period for reply except that is	one week						starting date is	(2	For any \
The period for reply fo	or weeks	[]		is	[]	ł	The	,	
The period for reply fo	or weeks	[]		is	[]	}	access dates are		
The following matters	will be included in the Early	v Warning	g Reg	gister			part of the Site		
For any Works instruc	ted under a Task Order, as	set out in	n the	relevant Task Ord	der		date		
Early warning meeting	gs are held at intervals no lo	onger tha	n		on	e month	l		
						•	Works invant Tas		ted under er
2 The Contrac	tor's main responsibilities	3							
If the <i>Client has</i> identi	fied work which is set to me	et a state	ed						
condition by a key data	te. The <i>key date</i> s and <i>condi</i>	itions to b	e me	et	L		1 For a	-	orks instruc Order

 $^{^4}$ This is likely to include the scope attached to the Task Order and all other relevant design documents where the works are full design and build (Part B Scope).

	nt of % per annum
5 Payment	Late 6 Compensati
The defect correction period for is wks	
The defect correction period for is wks	
The defect correction period is four weeks except that:	provisi ons of the
The period between Completion of the whole of the works and the wks	the time
The period after the Contract Date within which the <i>Contractor</i> is to wks policy statement and quality plan is	o submit a quality 2
4 Quality Management	The force interest from rate is, t unless ime to
The period after the Contract Date within which the <i>Contractor</i> is to programme for acceptance is	For any Works instructed under a Task Order, as set out in the relevant Task Order
The Client [is/is not] willing to take over the works before the	ar , month Englan d base rate in
The completion date for the whole of the works is	For any Works instructed under a Task Order, as set out in the relevant Task Order
The Contractor submits revised programmes at intervals no longer	
3 For any Works instructed under a Task Order, as set out in the relevant Task Order	The (Intere For any Worksbinstruste Aunder a Task Order, AB, Aet out 1984 he relevant Task Onder otherwi is se
	ercial Debts
	relevant Tasto Otraler of tis the Comm
2 For any Works instructed under a Task Order, as set out in the relevant Task Order	The above For any Wo rks rinestruttled under a Task Order, yas fatee ott ankthe

					_
The place where weather is to be recorded is	For any Works in Task Order	structed under a Task Order, as	set out in the	e relevant	
-			N/A		
The weather measurements to be	e recorded for eac	h calendar month are	(1)		¬ a
the cumulative rainfall (r	•	F mm	,		n
the number of days withthe number of days with		erature less than 0 degrees Cels	siu(2)	N/A	
 the number of days with 	snow lying at 9AM	√ hours GMT			С
			(3)	N/A	
and these measurements:			(5)	T	е
			O	h	а
N/A			7	e	g
			_	m	а
					_ i
The weather measurements are	supplied by	The Met Office			
				i	n
The weather data are the record	s of past weather i	measurements for		n	S
		For any Works instructed under	r o Took Or	dor on not	1
each calendar month which were	e recorded at	For any Works instructed unde out in the relevant Task Order	ra rask Ord	ier, as sei	
				i	t
					٦
and which are available from		For any Works instructed unde out in the relevant Task Order	r a Task Ord	der, as set	
				m	⊿
				u	0
Assumed values for the ten year		ta for each weather		m	S
measurement for each calendar	month are			a	S
				m	0
[]			7	0	f
L J			_	u	0
				G	Ū
The value engineering percentage here, i	ge is 50%, unless a	another percentage is stated	50		
% which case it is				_	-
				n	r
				t	d
				0	a
These are additional compensat	ion events			f	m
	ion ovonto			C	a T ~
				0	g
N1/A				V	e
• N/A				e	t
				r	0
				f	p
					⊢ r
O Habilitian and David				r i	0
8 Liabilities and insuran	ce			n	р
				n	e
These are additional Client's liab	oilities			S	r +
				u r	t
				r	У

	for bodily injury person (not an e connection with	ept the works, Plant and Materials and Equipment) and liability odily injury to or death of a on (not an employee of the Contractor) arising from or in ection with the Contractor Providing the Works for any one t is: £10,000,000.						
0	The minimum lir or bodily injury t employees of th their employment £10,000,000	o e Contr	actor arising	Addres s for electro	will choose an Adjudic ator if the			
0	The Contractor those insurance	s identii				nic commu nicatio ns	Parties cannot agree a choice	
Resolving and	avoiding disput	tes				Tl	is	
TI - (2) 12-		N1/A				The		
The <i>tribunal</i> is		N/A				Adiudio		
The arbitration	nrocoduro io	N/A				Adjudic		
The arbitration	procedure is	IN/A				ator is		
The place wher	o arbitration is to	ho holo	Lic		N/A	at01 15		
The place wher	e <i>arbitration</i> is to	be neic	1 15		IN/A	the	The	1
•	organisation who e <i>arbitration proc</i>				ties cannot agree an arbitrator is	person chosen by	Chairm an for the	
N/A						The	time being	
The <i>Senior Rep</i>	presentatives of th	ne <i>Clier</i>	nt are			Parties	of the Techno logy	
Name (1)		[]					
Address for con	nmunications							
Address for als		[]					
Address for elections								
Name (2)		[]					
		[]					Т
							and	

The Adjudicator nominating body is

The Technology and Construction Solicitors Association

Option X5: Sectional Completion Option X5 is used The completion date for each section of the works is required. description section Adva completion date nce (1) (2) (3) (4) payment An advance Option X7: Delay damages payment bond is/is not For any Works instructed under a Task Order, as set out in the relevant Task Order per day or **Option X13: Performance Bo** Delay damages for Completion of the whole of the works are part thereof If Option X13 is used The a is bond

Option X16: Retention Option

If Option X 14 is used is Thea

The po from v instalr

The in (eithei the pa

he Contr actor may/ may not

give the Client a retention bond. (Delete as Re	etention bond		
applicable).			Option Z: Additi onal
Option X18: Limitation of Liability			conditi
			ons of
			contra
Refer to the schedule of amendments contain Reference RM6088 and as attached at Apper within Part E of Appendix 1 to this Contact.			ct
Where the parties agree that Option X18 shall	l apply:		
The Contractor's liability for Defects under X1 respect of any one claim, to ⁵ :	8.4 is limited, in		
			The
			additio
The Contractor's liability under or in connection Contract under X18.5 is limited, in respect of contract under X18.5 is limited.			nal
series of claims arising from the same origina	ting or underlying		
cause, to:			
Option X20: Key Performance Indicators			
		_	contra
The incentive schedule for Key Performance	Indicators is in	Attached at App Delivery Agreen	ct are endix 8 of the
A report of performance against each Key Pe Indicator is provided at intervals of	rformance	Monthly	ng clause s
Option Y(UK)2: The Housing Grants, Constr	uction and Regene	eration Act 1996	
The first invoice date is	For any Works ins in the relevant Tas		ask Order, as set out
Later <i>invoice dates</i> occur on the same day of e <i>invoice date</i> until all amounts due to the <i>Contra</i> paid.			
5 5 Lot A: £1m, Lot B: £5m and			
Lot C:			

£10m.

PART TWO –	DATA PROVID	ED BY TH	IE CONTRACTO)R	۰	The Scope for the	For any Wor
Completion of the coreate a complete		ding to the	e Options choser	n, is essential t	o	Contr actor' s desig n is in	ks instr ucte d und
1 General							er a Tas k
The Contractor is							The program
Name		[]			out	me identifie d in the
Address for commu	unications	[1			the Ca pit al	Contract Data is
Address for electro communications	nic	1	1			ork s/L ife	the
The fee percentage	e is %]6				Te nd er Ra tes an d Pri	
For any V	Works instructed	under a T	ask Order, as se	t out in the rela	evant	ce	For
The working areas		unaci a i	ask Oraci, as sc	t out in the ren	Svarit	s (as	
g		ask Order				ap pro pri	Works
The key persons a	re those identifie	d in the <i>ke</i>	ey persons sched	<i>dule</i> in	under	a Task C	instructed Order, as set ant Task
The following matte	ers will be include	ed in the E	early Warning Re	gister). ⁷ Thi	
For any Works inst	ructed under a T	ask Order	r, as set out in the	e relevant Tas	k Order	s will be the Pa rt B Sc op	er a Tas k The
2 The Contr	actor's main re	sponsibili	ities			e.	for the whole of
		-				3	the Time s is
							Order, as set

		Order		
Payment				The percent age for adjust
he <i>activity schedule</i> is in			For any Works instruction order, as set out in the Order	
the tendered total of the Prices is			For any Works instruction order, as set out in the	
			Order	Equip
Resolving and Avoiding Disputes				ment in he people rates are
The Senior Representatives of the Contrac	ctor are		C	eategory of person ed list
Name (1)	[1		(state plus or is
value (1)	[]		minus)
Address for communications				The rates
Address for electronic communications	[]		for other Equip
				ment are
Name (2)]]	p e	oublish Equipm
Address for communications	[]	n	Equip rate
Address for electronic communications			е	edition
	[1	а	at the Contac
		. 6		Date of the st
Data for the Short Schedule of Cost Cor	nponer	nts°	p	publish ed by

 $^{^{\}rm 6}$ To be completed using either the Tender Schedule of Cost Components or, if no schedule is submitted by the Contractor, the rates in the CCS framework.

The rates for Defined Cost of manufacture and fabrication outside the Workings Areas by the <i>Contractor</i> are	<u>А</u> р	
category of person rate	<u>p</u>	
	<u>е</u> <u>n</u>	
	<u>d</u>	
	<u>i</u> <u>x</u>	
	<u>4</u>	
		<u>Р</u> а
		<u>r</u> <u>t</u>
		<u>A</u> -
The rates for Defined Cost of design outside the Workings Areas are		M
category of person rate		<u>a</u>
		<u>n</u> a
		g
		<u>e</u> m
		<u>е</u>
		<u>'</u>
		<u>t</u> S e
		<u>r</u>
		<u>v</u> <u>i</u>
The categories of design people whose travelling expenses to and from the Working Areas are included in Defined Cost are		<u>I</u> C
Working / Hode are moladed in Bollinea Cook are		<u>e</u> <u>s</u>
		<u>s</u>
	1	
	<u>Workin</u> g	
	ਖ਼ togethe	
Appendix 3	ī	
NUTS Codes LOT C South		1
NUTS Codes – LOT C, South H1 H2 H3 I3 I4 I5 I6 I7 J1 J2 J3 J4 K1 K2 K3 K4 L	1 L2	

tion to the carrying out of the Works and any obligations under this Contract, the Parties shall work together in order to achieve the Objectives. To that end the Client and the Contractor agree they shall each give to, and welcome from, the other, and the other Project Participants, feedback on performance and shall draw each other's attention to any difficulties and shall share information openly, at the earliest practicable time. They shall support collaborative behaviour and address behaviour that does not comply with the Objectives. During the Delivery Term the Contractor shall:

W

i

m

е

е

t

а

t

r

е

g

u

а

r

n

t

е

r

٧

а

ı

s

а

n

d

0

n

t

h

е

f

i

r

S

t

а

n

n

i

٧

е

r

S

а

r

У 0

f

t

h

е

D

е

i

V

е

r

У Τ

е r

m

t

0

s

h

а

r

е i

n

f

0

r

m

а

t

i

0

n

r

е

ı

а

t

n

g

t

0

t

h

е

Ρ

r

0

е

С

t

а

- 1.2
 - collaborate with the Client and the Project 1.2.1 Team to achieve the Objectives;
 - 1.2.2 work with the Client in a supportive manner;
 - 1.2.3 produce such reports and documentation as may reasonably be requested from time to time by the Client:
 - 1.2.4 work collectively with the Client and the Project Team to support the delivery of the Works on a best for project basis;
 - 1.2.5 work collectively with the Client and the Project Team to discuss the Pipeline Programme and whether certain works and/or packages should be bundled into appropriate Task Orders;
 - 1.2.6 develop and use common systems and processes with the Client and the Project Participants;
 - 1.2.7 discuss costings of the Works on an open-book

basis;

1.2.8 work with the Client to determine whether the

Contractor should

select additional NUTS Codes to those NUTS Codes initially selected by the Contractor;

- 1.2.9 work with the Client to agree acceptable processes for demonstrating value for money and understand the key drivers for value for money; and
- 1.2.10 give advice, information and opinion fully, openly and objectively to the Client and the Project Team.
- 1.3 The role of the Project Team is to guide the successful delivery of the Project and the Works. The Project Team

nd the Works, to consider the risks and opportunities affecting the Project and the Works, to consider how best to coordinate and manage the Project and the Works, to review progress and to make any decisions necessary for the successful delivery of the Project and the Works which are then communicated to all relevant Project Participants.

Ρ

ı

а

n

f

0

t

h

е

С

Ι

е

n

t

s

а

р

р

r

0

٧

а

Ι

Т

h

е

C

0

n

t

i

n

u o

u

s

Ι

m

р

r

0

٧

е

m

е

n t Ρ

а

n

s

h

а

Ι

b

е

i

n

t

h

е

f

0

r

m

а

t

а

g

r

е

е

d

b

е

t w

е

е

n

t

h

е

Р

а

r

t

i

е

s

а

n d

s

h

а

ı

- 1.4 In the event of any dispute between the Parties and/or the members of the Project Team, it is the intention that any court or adjudicator or other forum to which the dispute is referred shall take account of the Objectives and of the Parties' adherence to it when making any award.
- 1.5 The Client and the Contractor shall co-operate with the other members of the Project Team in the provision of information with a view to ensuring that relevant information is provided to all Project Participants needing this information in a timely fashion.
- 1.6 Subject always to the terms of any Task Order, if the Contractor identifies any ambiguities or discrepancies in any information provided by the Client or any other member of the Project Team, the Contractor shall agree how to resolve them after consultation, if appropriate, with the Project Team. If no agreement is reached, the Client shall issue an instruction.
- 1.7 The Client hereby elects to be treated as the only client in respect of the Works for the purposes of the CDM Regulations.
- 1.8 The Contractor shall act as "principal contractor" in respect of the Works for the purposes of the CDM Regulations.

2 Continuous Improvement

- 2.1 The Contractor shall, throughout the Delivery Term, adopt a policy of continuous improvement in relation to the Works pursuant to which it will regularly review with the Client the Works and the manner in which it is providing the Works with a view to reducing the Client's costs (including the Prices and/or the Price List) and/or improving the quality and efficiency of the Works. The Contractor and the Client will provide to each other any information which may be relevant to assisting in fulfilling these objectives.
- 2.2 Without limiting paragraph 2.1 and if the Client so requires at any point during the Delivery Term, the Contractor shall produce a plan for improving the provision of Works and/or reducing the Prices and/or the Price List (without adversely affecting the performance of this Contract and/or any Task Order) ("Continuous Improvement

S

0

Ι

u

t

i

0

n

а

С

С

е

p t

а

b

Ι

е

t

O

t

h

е

C

Ι

i

е

n

t2

а

n

d

t h

е

C

0

n

t

r

а

С

t

0

r

t

h

е

р

r

о С р

r

0

е

s

s

u

s

е

d

i

n

W

2

contain, as a minimum, proposals in respect of the following:

- 2.2.1 identifying the emergence of relevant new and evolving technologies;
- 2.2.2 changes in business processes of the Contractor or the Client and ways of working that would provide cost savings and/or enhanced benefits to the Client (such as methods of interaction, supply chain efficiencies, reduction in energy consumption and methods of sale);
- 2.2.3 new or potential improvements to the provision of the Works including the quality, responsiveness, procedures, benchmarking methods, likely performance mechanisms and Client support services in relation to the Works; and
- 2.2.4 measuring and reducing the sustainability impacts of the Contractor's operations and supply-chains pertaining to the Works and identifying opportunities to assist the Client in meeting their sustainability objectives.
- 2.3 The initial Continuous Improvement Plan for the first Service Quarter is attached at Part B of this Appendix 4. As and when required, but at least once each Service Quarter, the Contractor shall review the Continuous Improvement Plan and submit any updates or changes to the Client for the Client's approval.
- 2.4 The Client shall notify the Contractor of its approval or rejection of the proposed Continuous Improvement Plan or any updates to it within twenty (20) Business Days of receipt. If it is rejected then the Contractor shall, within ten (10) Business Days of receipt of notice of rejection, submit a revised Continuous Improvement Plan reflecting the changes required. If this revised Continuous Improvement Plan is also rejected the Contractor shall provide a second revised Continuous Improvement Plan for approval. Should this second revised Continuous Improvement Plan be rejected ("Second Rejection") the Parties may escalate this to the Senior Representatives who shall meet to discuss the Second Rejection within ten (10) Business Days of the notice of rejection. Where such meeting achieves a solution acceptable to the Client and the Contractor, such solution shall be implemented by the Client and the Contractor. Where such meeting does not achieve a

- ement Plan is approved, it shall constitute the Continuous Improvement Plan for the purposes of this Contract.
- 2.6 The Contractor shall provide sufficient information with each suggested improvement to enable a decision on whether to implement it. The Contractor shall provide any further information as requested.
- 2.7 All costs relating to the compilation or updating of the Continuous Improvement Plan and the costs arising from any improvement made pursuant to it and the costs of implementing any improvement, shall have no effect on and are included in the Prices and/or the Price List.
- 2.8 Should the Contractor's costs in providing the Works to the Client be reduced as a result of any changes implemented, all of the cost savings shall be passed on to the Client by way of a consequential and immediate reduction in the Prices and/or the Price List.

3 Allocation of risks

- 3.1 The Project Team shall carry out, complete and populate the Project Risk Register to identify:
 - 3.1.1 potential risks relating to the delivery of the Project, the occurrence of which are capable of adversely affecting the time for completion, cost or quality of the Project;
 - 3.1.2 the probability of these risks occurring;
 - 3.1.3 a financial estimate of the most likely consequences of each risk occurring; and
 - 3.1.4 the actions agreed to be taken to mitigate, manage or remove each risk and the individual responsible for reporting on the progress of the actions agreed.
- 3.2 The Project Team shall regularly update and amend the Project Risk Register (seeking the assistance of other Project Participants as considered appropriate) and shall arrange regular meetings with the Client and members of the Project Team to review and update the Project Risk Register and to consider:
 - 3.2.1 any new risks that have arisen since the date of the

last review;

3.2.2 the steps taken to prevent/mitigate previously identified risks:

3.2.4 the prioritisation of all continuing risks and agreement of an action plan in respect of, and risk owners for, all risks prioritised as serious risks.

4 Problem solving

- 4.1 Without prejudice to option W2, as soon as a Party is aware of any difference or dispute with the Client and the Contractor arising out of or in connection with the Works (a "Difference") such Party will give notice to the other.
- 4.2 Upon receipt of the notice in paragraph 4.1, there shall be a meeting between respective representatives of the Client and the Contractor having authority to settle the matter. Where such meeting achieves a solution acceptable to the Client and the Contractor, such solution shall be implemented by the Client and the Contractor.
- 4.3 If such meeting fails to achieve a solution acceptable to the Client and the Contractor, the Difference shall be escalated to the Senior Representatives and within ten (10) Business Days of the meeting referred to in paragraph 7.2 the Senior Representatives shall meet to discuss the Difference. Where such meeting achieves a solution acceptable to the Client and the Contractor, such solution shall be implemented by the Client and the Contractor. Where such meeting does not achieve a solution acceptable to the Client and the Contractor, the process used in W2 shall be followed.

5 Fee

5.1 The Contractor agrees that the fee for the Management Services shall be nil.

Part B - Continuous Improvement Plan for the first Service Quarter

As & when requested by client

Appendix 5

Rates and Prices

REDACTED

1 Rates and Prices where is no analogous rate

- 1.1 Where the Contractor has, in its Task Order Proposal for a relevant Proposed Task, identified that there are no analogous Rates and Prices for any element of the Works, then the Parties shall seek to agree new Rates and Prices as follows:
 - 1.1.1 the Parties shall, within five (5) Business Days of receipt by the Client of the relevant Task Order Proposal, meet to discuss whether there is any sufficiently detailed cost element within the Price List to derive new Rates and Prices for the whole or the relevant part of the Works;
 - 1.1.2 where the Parties:
 - (i) agree that there is no sufficiently detailed cost element contained within the Price List to derive new Rates and Prices; or
 - (ii) fail to agree a new set out Rates and Prices,

then the provisions of paragraph 2 shall apply to determine the applicable price(s) for the whole or the relevant part of such relevant Works; and

1.1.3 any new Rates and Prices agreed or determined pursuant to this Appendix shall:

- (i) include pricing for all risks associated with the relevant Works; and
- (ii) exclude any costs in respect of which the Contractor is entitled to recover under any other provision of this Contract.
- 1.2 Where any new Rates and Prices are agreed or determined pursuant to this Appendix, such new Rates and Prices and shall be incorporated into the Price List and shall apply to the calculation of the costs for the whole or the relevant part (as the case may be) of the relevant Works for all subsequent Task Order Proposals.
- 1.3 Where paragraph 1.1.2 applies then the Parties may either:
 - 1.3.1 agree to refer the matter for determination by the Adjudicator pursuant option W2; or
 - 1.3.2 follow the competitive tendering procedure set out in paragraph 3, in either case to set the new Rates and Prices for the whole or the relevant part of the relevant Works.
- 2 <u>Indexation of Life Cycle Tender Rates and Prices and SSCC Tender Rates and Prices</u>
 - 2.1 Subject to paragraphs 2.3, 2.8 (inclusive) on each date which is 1 November that occurs during the Delivery Term ("Framework Anniversary"), either party may apply to change the applicable SSCC Tender Rates and Prices by multiplying the relevant rates and prices by the percentage changes applicable
 - to the CCS Framework Rates and Prices ("Framework Index") published since the date of this Contract. (BCIS CPI Index used).
 - 2.2 Subject to paragraphs 2.3, 2.8 (inclusive), on each anniversary of the date of this Contract ("Delivery Anniversary"), either party may apply to change the applicable Life Cycle Tender Rates and Prices by multiplying the relevant figures by the percentage changes in the BCIS All-in Tender Price Index (TPI) ("Delivery Index") published since the date of this Contract.
 - 2.3 If the Framework Index and/or the Delivery Index has not been published for the relevant month in which the Framework Anniversary or Delivery Anniversary (as appropriate) occurs, then the last published value of that index available at the relevant anniversary shall be used.
 - 2.4 Where the Framework Index and/or the Delivery Index is no longer published, the Client and Contractor shall agree a fair and reasonable replacement that will have substantially the same effect.
 - 2.5 Any change agreed in accordance with paragraphs 2.1 2.4 (inclusive) shall apply to any subsequent Task Orders which are issued after any such change has been agreed.
 - 2.6 Any Life Cycle Tender Rates and Prices which are identified in the Life Cycle Price List as being Rates and Prices that are fixed ("Fixed Rates and Prices") are

- not capable of being changed and any change agreed to the Life Cycle Tender Rates and Prices shall not apply to any such Fixed Rates and Prices.
- 2.7 Any uplift agreed in accordance with paragraphs 2.1 2.4 (inclusive) shall not exceed the applicable CCS Framework Rates and Prices.
- 2.8 Paragraphs 2.1 2.4 (inclusive) shall not apply to:
 - 2.8.1 any percentages referred to within the Life Cycle Tender Rates and Prices;
 - 2.8.2 any percentage set out and/or referred to in the Short Schedule of Cost Components; and/or
 - 2.8.3 the Capital Works Tender Rates and Prices,

and, for the avoidance of doubt, there shall be no change or uplift to any such rates and prices identified in paragraphs 2.8.1 - 2.8.3 (inclusive) of this paragraph 2.

3 Procedure

- 3.1 The following principles are to govern each tendering exercise initiated pursuant to this paragraph 3:
 - 3.1.1 the object of the tender is to obtain open-market rates and prices;
 - 3.1.2 the process is to be genuinely competitive;
 - 3.1.3 the evaluation of tenders is to be fair, robust and transparent; and
 - 3.1.4 tendered prices are to be presented in a form that demonstrates value for money for the Works to which they relate.
- 3.2 The Contractor shall be responsible for the carrying out of the tendering procedure in accordance with this paragraph 3.
- 3.3 The Contractor shall procure that tenderers will be invited by open invitation or from no fewer than three (3) suitably qualified, experienced and willing contractors.
- 3.4 The Contractor shall discuss and agree with the Client those contractors who it proposes to invite to tender and the Client may require other potential contractors to be invited to tender as part of such discussions.
- 3.5 The Contractor will provide the Client with all relevant background information on the tenderers it proposes pursuant to paragraph 3.4, in order to demonstrate such tenderers' capability (including technical and financial capability) and capacity to undertake the relevant work or supply.
- 3.6 The Contractor shall:
 - 3.6.1 prepare all necessary tender documentation;
 - 3.6.2 provide copies to the Client for review and comment before

- despatch, allowing the Client a reasonable time for review prior to such proposed despatch; and
- 3.6.3 where applicable, revise such tender documentation to take into account the Client's comments pursuant to paragraph 3.6.2.

3.7 The Contractor shall:

- 3.7.1 prepare a tender evaluation methodology prior to the tender issue date and issue such methodology to the Client for review and comment, allowing the Client a reasonable time for review prior to such proposed despatch; and
- 3.7.2 where applicable, revise such evaluation methodology to take into account the Client's comments pursuant to paragraph 3.7.1.
- 3.8 The Contractor shall ensure that all prices submitted as part of any tender submission will be calculated (to the extent practicable) on the same basis and using the same assumptions as applied to derive the Unit Rates.
- 3.9 The Contractor shall invite the Client to attend at all tender openings and any subsequent interviews, value engineering workshops or other meetings with all or any (as applicable) of the tenderers.
- 3.10 The Contractor shall provide copies of all issued tenders and submissions received from tenderers, together with copies of all relevant supporting information, minutes of meetings and reports in respect of each tender, (such copies to be delivered to the Client as soon as possible from the date the Contractor produced, issued or received the same (as the case may be)).
- 3.11 The Contractor shall prepare a report on each submitted tender response, disclosing the basis upon which the evaluation was carried out (including details of scoring), and including a recommendation as to the tender which represents, having regard to the relevant agreed tender evaluation criteria, fair open market rates and which, in the Contractor's opinion offers value for money and why it should be utilised in order to determine the relevant Rates and Prices for the relevant Works for the purposes of this Appendix.
- 3.12 The Contractor and the Client shall discuss such tender report and shall agree which tender should be utilised in order to determine the relevant Rates and Prices for the relevant Works for the purposes of this Appendix.
- 3.13 If the Contractor and the Client cannot agree which tender should be utilised in order to determine the relevant Rates and Prices for the relevant Works, the matter may be referred by either party to the dispute resolution procedure set out in option W2.
- 3.14 Following agreement to a tender by the Contractor and the Client (as referred to in paragraph 3.12) or, if paragraph 3.13 applies, upon determination pursuant to the dispute resolution procedure, the relevant Rates and Prices for the relevant Works shall become the relevant Rates and Prices for the purposes of this Appendix.

- 3.15 Without prejudice to any other provision of this Appendix, the Contractor shall provide to the Client:
 - 3.15.1 market advertisements proposed for any tender exercise;
 - 3.15.2 technical specifications relevant to any tender;
 - 3.15.3 copies of all documentation and communications submitted by a tenderer relating to a tender; and
 - 3.15.4 the tender evaluation records and reports prepared by the Contractor,

in each case as soon as reasonably practicable from the date the Contractor has produced, issued or received the same.

- 3.16 The Client may enter onto any premises of the Contractor (which any of them own/use for the purposes of their businesses) at all reasonable times to observe, inspect and satisfy itself as to the adequacy of the carrying out of any tender in accordance with the procedure set out in this Appendix.
- 3.17 The Contractor shall indemnify and keep the Client fully indemnified at all times from and against all claims, demands or notices which may be brought or alleged or threatened against the Client and from and against all losses and fines which the Client may suffer or incur in relation to any such claims, demands or notices which occur as a result of a breach of this Appendix and/or any claim made by any person (including, without limitation, a tenderer or prospective tenderer) that is not awarded a contract pursuant to any tender governed by the procedure set out in this Appendix.

4 General

- 4.1 The Rates and Prices are deemed to include all costs, disbursements, expenses, required accommodation and accommodation costs, personnel rates and charges (including all costs associated with their employer) and other items necessary for the Contractor to fulfil all its obligations and liabilities under or referred to in any Task Order.
- 4.2 The Rates and Prices shall specify any key personnel and the parties agree that no other persons will be allowed in the establishment of prices under a Task Order unless specifically requested or otherwise agreed by the Client.
- 4.3 In relation to the Contractor's equipment costs, the Rates and Prices shall specify the accommodation required by the Contractor and the rates for any such accommodation and the parties agree that no other accommodation and/or rates will be allowed in the establishment of prices under a Task Order unless specifically requested or otherwise agreed by the Client.
- 4.4 The parties agree that the Rates and Prices shall not include any mechanism or rates or prices in relation to:
 - 4.4.1 external scaffolding; and/or

- 4.4.2 external access equipment (including but not limited to, scaffolding, ladders, tower scaffolds, access platforms, tracked spiders and electric hoists).
- 4.5 Notwithstanding that any information described in paragraph 4.4 is set out in the Rates and Prices, in breach of paragraph 4.4, the parties agree that this information is to be disregarded for the purposes of establishing the rates and prices for:
 - 4.5.1 external scaffolding; and/or
 - 4.5.2 access information equipment (including but not limited to, scaffolding, ladders, tower scaffolds, access platforms, tracked spiders and electric hoists),

and if the Client or the Project Manager specifies that a Task Order requires any equipment identified in paragraphs 4.5.1 and/or 4.5.2 the Contractor shall follow the process set out in paragraph 3 of this Appendix 5 to obtain, from at least three specialist contractors and/or suppliers, a competitively tendered price for the erection, installation, hire, use, striking and/or dismantling of any such access information equipment.

4.6 The data used for the Short Schedule of Cost Components under any Task Order shall incorporate the applicable SSCC Tender Rates and Prices or, if the Contractor has not provided such rates and prices, then the CCS Framework Rates and Prices shall apply.

Appendix 6

Schedule of Amendments

STANDARD 'BOILERPLATE' AMENDMENTS

NEC4

JANUARY 31, 2019

CABINET OFFICE

amendments project addresses a need to simplify the inclusion of government-specific clauses to the NEC, JCT and PPC2000 contracts. Centrally mandated government policies and some legislative requirements were being applied by a range of government departments, but as separate operations and with differing approaches. Scope was identified for a simple and standard set of terms which provide a unified front to implement policy and reduce the need for excessive additional drafting, creating a more efficient standardised approach. These terms would be applied across government construction contracts.

In order to bring about this situation, a cross-governmental review of construction contract amendments was undertaken by the Crown Commercial Service (CCS) and the Infrastructure and Projects Authority (IPA). Eighteen clauses were identified as those which would benefit most from the standardisation described above. These clauses were reviewed and redrafted to enhance their ease of comprehension, with the core wording translated to NEC, JCT and PPC2000 terminology.

These eighteen clauses are replicated within the NEC, JCT and PPC2000 boilerplate documents. This is the **NEC4 version.**

The clauses should be **unamended** save for those instances with an additional guidance note. Not all will be relevant to each project, and additional clauses may be required where not covered by this document. Those 'boilerplate' clauses not required can be removed and additional, project specific clauses may be added.

Process

The clauses are amended to the contract by way of an additional Schedule of Amendments. This must be referred to in the base contract. The following segment indicates the modification which must be made to the base contract, as well as the steps needed to incorporate the Boilerplate Amendments.

NEC4 Engineering & Construction Contract

• In Contract Data Part One, complete the clause headed "If Option Z is used" to read:

The *additional conditions of contract* are as detailed in the appended Schedule of Amendments which is to be read and construed accordingly.

- Append pages 8 to 42 of this Standard 'Boilerplate' Amendments document to the standard contract document as this Schedule of Amendments.
- Remove or strikethrough those clauses which do not apply to the current project and amend terms as provided by guidance notes. Add additional, project specific amendments in the normal way.

Table of Contents

	Summary of Clauses4	
N	NEC 4	
	3	
	Z2 Definitions	
	Z4 Admittance to Site	
	Z5 Prevention of Fraud and Bribery	12
	Z7 Legislation and Official Secrets	13
	Z10 Freedom of Information	14
	Z13 Confidentiality and Information Sharing	15
	Z14 Security Requirements	17
	Z16 Tax Compliance	7
	Z22 Fair Payment 1	18
	Z42 The Housing Grants, Construction and Regeneration Act 1996	18
	Z55 Intellectual Property Rights	18
	Z46 MOD DEFCONS	20
	Z47 Small and Medium Enterprises (SMEs)	20
	Z48 Apprenticeships	
	Z100 GDPR	
	Z101 Cyber Essentials	

Schedules

GDPR	
23	
Security 30	Provisions
Cyber 41	Essentials

SUMMARY OF CLAUSES

The following descriptions are of all the clauses addressed by the Standard 'Boilerplate' Amendments project. It should be noted that some of the clauses differ from document to document.

1. Definitions

A list of additional definitions must be included as an amendment to help explain the meaning of subsequent Boilerplate clauses.

2. Admittance to Site

This clause specifies additional provisions around how individual people may be admitted to the site, and the considerations which must be taken. This includes (but is not limited to) the provision of a list of employee names, obligations as to security passes, and the prevention of unauthorised access or taking of photographs.

3. Prevention of Fraud and Bribery

The Boilerplate clause expands the coverage of the standard contract Fraud and Bribery provisions. It introduces a 'Prohibited Act', also defined in the Boilerplate document, which must not be committed and which must be subject to suitable caution and management. The Contractor must hold subcontractors to the same standards, keep appropriate records of compliance, and immediately notify the Client of potential breaches and work with them to rectify the situation.

4. Official Secrets Act

Contractors are often required to abide by this Act due to the sensitive nature of some public sector projects. The Boilerplate clause saves Clients from drafting this themselves if

required, creating an obligation to comply with this Act and, where appropriate, section 11 of the Atomic Energy Act 1946.

5. Freedom of Information

As government departments are usually required to comply with Freedom of Information Act requests, extra clauses detailing how this obligation is to be respected must be included. The Boilerplate clause obliges the Contractor to work with the Client in satisfying these requests in certain ways. Among other considerations, this involves the retention and transferral of relevant information, communicating requests for information to the Client in a timely manner, and generally helping the Client in responding to the request.

6. Confidentiality and Information Sharing

Some public sector information is sensitive and cannot be shared, while at other times organisation must share details about its processes in the interest of transparency. As such, this clause provides obligations for both parties to safeguard confidential information, exceptions where that obligation does not apply, and additional restrictions on the Contractor and further rights for the Client.

7. Security Requirements

This clause is a preface to a schedule requiring the Contractor to create and maintain a comprehensive Information Security Management System. This must be agreed with the Client, contain measures sufficient to ensure security on the project in question, and be regularly reviewed to reflect changes in good practice or project details. It must be tested appropriately and be fully compliant with ISO 27001, subject to audits as required. The schedule also indicates some of the steps to be taken in the event of a security breach.

8. Tax Compliance

With the inclusion of this clause, the Contractor is under an obligation to notify the Client of relevant Tax Non Compliance. The Contractor must provide more information if the Occasion of Tax Non Compliance occurs prior to the end of the defects correction period (NEC) / Rectification Period (JCT and PPC).

9. Contract (Rights of Third Parties) Act 1999

Excluding third party rights is a common clause in all manner of contracts. The Boilerplate clause removes that exclusion in the case of collateral warranties – a common and often necessary provision in public sector construction. It should be noted that this does not apply to the NEC4 Boilerplate document as the contract directly deals with this under Option X8.

10. Fair Payment

This is a clause also aimed at improving how subcontractors are paid, similarly endorsed in the Government Construction Strategy 2016. Obligations are placed on the Contractor to assess and promptly pay subcontractors, and to ensure that these obligations are also included in their contracts with subcontractors. It should be noted that Fair Payment is a separate clause with the NEC Boilerplate document, whereas within JCT and PPC it is combined with the SME provisions to form 'Conditions of Sub-Contracting' (JCT) or 'Supply Chain' (PPC).

11. Building Information Modelling (BIM)

Promoting and spreading the use of BIM techniques is a major government construction objective, as identified in the three main policy documents – the Government Construction Strategy 2016, Construction 2025, and the Construction Sector Deal. It has been mandated for all central government departments and is aimed at enhancing efficiency and reducing costs across the industry. This clause provides a mechanism for BIM Protocols to be applied as indicated in the Employer's Information Requirements, as well as an option to incorporate a specific type of Protocol, namely the CIC (Construction Industry Council) BIM Protocol. This clause is not replicated in NEC4, which has overlapping mechanisms with the Boilerplate BIM Provision.

12. The Housing Grants, Construction and Regeneration Act 1996 ('Construction Act 2011')

This is an NEC-only clause which expands on an existing provision. If NEC Option Y(UK)2 applies, then the Construction Act also applies to this contract even if the project is in Northern Ireland.

13. Intellectual Property Rights

This indicates that the Contractor provides to the Client an irrevocable, royalty free and nonexclusive licence to use the Intellectual Property of the Contractor. The Client may transfer these rights in a variety of circumstances, and the Contractor is subject to a number of additional obligations.

14. MOD DEFCONs

This provision is applicable only to Ministry of Defence projects and contracts. It incorporates their special terms and conditions.

15. Small and Medium Enterprises (SMEs)

Government policy dictates that SMEs should be encouraged and brought into public sector projects, as reinforced in the Government Construction Strategy 2016, Construction Sector Deal and Construction 2025. There is a general target for 33% of central government procurement spend going to SMEs by 2022. This Boilerplate clause requires Contractors to employ a certain amount of SMEs as subcontractors, and to respect a number of other obligations regarding reporting and how they manage these SMEs.

16. Apprenticeships

In a similar way to SMEs, there is an overarching government policy for public sector organisations to promote the creation and use of apprenticeship schemes, as per the Government Construction Strategy 2016, the Construction Sector Deal and Construction 2025. In particular, a 2015 Procurement Policy Note describes the steps that public sector organisations must take to ensure they are meeting the government's apprentice aims. This Boilerplate provides a way for Clients to ensure that Contractors do this by creating an obligation to employ certain amounts of apprentices. They must also provide further training opportunities and information about the Government Apprenticeship programme, and engage with the Project Manager to review and discuss a number of measures relating to Apprenticeships.

17. GDPR

With the recent advent of the General Data Protection Regulation, every construction project is required to include provisions within their contracts to ensure compliance. The Boilerplate document includes a Schedule so these regulations can be complied with, with areas for the parties to fill in to reflect project specific data protection requirements.

18. Cyber Essentials

This clause provides a way to include the Government Cyber Essentials scheme into construction projects. This scheme provides for a number of controls which organisations should implement to reduce the risk of common internet based threats. The clause lists obligations on the Contractor to provide proof of the required certification at certain stages of the project, and to apply the same obligations to its sub-Contractors.

19. Project Bank Accounts

The Project Bank Accounts scheme is a government policy aimed at enhancing the speed with which payment progresses down the construction supply chain. The scheme has been promoted in the Government Construction Strategies and should be used within central government projects unless there are compelling reasons not to do so. Amended provisions have not been included within this boilerplate document, however their use is encouraged. They should be incorporated using the standard facilities within NEC, JCT and PPC documents.

SCHEDULE OF AMENDMENTS TO NEC4 ENGINEERING

AND CONSTRUCTION CONTRACT

Option Z2 - Identified and defined terms

Insert new clause 11.3 additional defined terms.

- 11.3 (1) Client Confidential Information is all Personal Data and any information, however it is conveyed, that relates to the business, affairs, developments, trade secrets, knowhow, personnel, and contractors of the *Client*, including all IPRs, together with all information derived from any of the above, and any other information clearly designated as being confidential (whether or not it is marked "confidential") or which ought reasonably be considered to be confidential.
- 11.3 (2) Client Data is the data, text, drawings, diagrams, images or sounds (together with any database made up of any of these) which are embodied in any electronic, magnetic, optical or tangible media, and
- which are supplied to the Contractor by or on behalf of the Client,
- which the Contractor is required to generate, process, store or transmit pursuant to this contract or
- which are any Personal Data for which the *Client* is the Data Controller to the extent that such Personal Data is held or processed by the Contractor.
- 11 (3) Commercially Sensitive Information is the information agreed between the Parties (if any) comprising the information of a commercially sensitive nature relating to the *Contractor*, the charges for the works, its IPR or its business or which the *Contractor* has indicated to the *Client* that, if disclosed by the *Client*, would cause the *Contractor* significant commercial disadvantage or material financial loss.
- 11.3 (4) Confidential Information is the Client's Confidential Information and/or the Contractor's Confidential Information.
- 11.3 (5) Contracting Body is any Contracting Body as defined in Regulation 5(2) of the

Public Contracts (Works, Service and Supply) (Amendment) Regulations 2000 other than the Client.

- 11.3 (6) Contractor's Confidential Information is any information, however it is conveyed, that relates to the business, affairs, developments, trade secrets, know-how, personnel and contractors of the *Contractor*, including IPRs, together with all information derived from the above, and any other information clearly designated as being confidential (whether or not it is marked as "confidential") or which ought reasonably to be considered to be confidential, including the Commercially Sensitive Information.
- 11.3 (7) Crown Body is any department, office or agency of the Crown.

- 11.3 (8) Data Controller has the meaning given to it in the Data Protection Act 2018.
- 11.3 (9) DOTAS is the Disclosure of Tax avoidance Schemes rules which require a promoter of tax schemes to tell HM Revenue & Customs of any specified notable arrangements or proposals and to provide prescribed information on those arrangements or proposals within set time limits as contained in Part 7 of the Finance Act 2004 and in secondary legislation made under vires contained in Part 7 of the Finance Act 2004 and as extended to National Insurance Contributions by the National Insurance Contributions (Application of Part 7 of the Finance Act 2004) Regulations 2012, SI 2012/1868 made under s.132A Social Security Administration Act 1992.
- 11.3 (10) Environmental Information Regulations is the Environmental Information Regulations 2004 and any guidance and/or codes of practice issued by the Information Commissioner in relation to such regulations.
- 11. 3(11) FOIA is the Freedom of Information Act 2000 and any subordinate legislation made under this Act from time to time together with any guidance and/or codes of practice issued by the Information Commissioner in relation to such legislation.
 - 11.3 (12) General Anti-Abuse Rule is
 - the legislation in Part 5 of the Finance Act 2013 and
 - any future legislation introduced into parliament to counteract tax advantages arising from abusive arrangements and to avoid national insurance contributions.
 - 11.3 (13) Halifax Abuse Principle is the principle explained in the CJEU Case C-255/02 Halifax and others.
 - 11.3 (14) Intellectual Property Rights or "IPRs" is
 - copyright, rights related to or affording protection similar to copyright, rights in databases, patents and rights in inventions, semi-conductor topography rights, trade marks, rights in internet domain names and website addresses and other rights in trade names, designs, Know-How, trade secrets and other rights in Confidential Information,
 - applications for registration, and the right to apply for registration, for any of the rights listed in the first bullet point that are capable of being registered in any country or jurisdiction,
 - all other rights having equivalent or similar effect in any country or jurisdiction and
 - all or any goodwill relating or attached thereto.
 - 11.3 (15) Law is any law, statute, subordinate legislation within the meaning of section 21(1) of the Interpretation Act 1978, bye-law, enforceable right within the meaning of section 2 of the European Communities Act 1972, regulation, order, mandatory guidance or code of practice, judgment of a relevant court of law, or directives or requirements of any regulatory body with which the *Contractor* is bound to comply under the *law of the contract*.
 - 11.3(16) An Occasion of Tax Non-Compliance is

- where any tax return of the Contractor submitted to a Relevant Tax Authority on or after 1 October 2012 is found on or after 1 April 2013 to be incorrect as a result of
- a Relevant Tax Authority successfully challenging the Contractor under the General Anti-Abuse Rule or the Halifax Abuse Principle or under any tax rules or legislation that have an effect equivalent or similar to the General Anti-Abuse Rule or the Halifax Abuse Principle or
- the failure of an avoidance scheme which the Contractor was involved in, and which was, or should have been, notified to a Relevant Tax Authority under DOTAS or any equivalent or similar regime and

where any tax return of the *Contractor* submitted to a Relevant Tax Authority on or after 1 October 2012 gives rise, on or after 1 April 2013, to a criminal conviction in any jurisdiction for tax related offences which is not spent at the Contract Date or to a civil penalty for fraud or evasion.

11. 3(17) Personal Data has the meaning given to it in the Data Protection Act 2018.

11.3 (18) Prohibited Act is

- to directly or indirectly offer, promise or give any person working for or engaged by the
 Client or other Contracting Body or any other public body a financial or other
 advantage to
- induce that person to perform improperly a relevant function or activity or
- reward that person for improper performance of a relevant function or activity,
- to directly or indirectly request, agree to receive or accept any financial or other advantage as an inducement or a reward for improper performance of a relevant function or activity in connection with this contract,
- committing any offence
- under the Bribery Act 2010 (or any legislation repealed or revoked by such Act),
- under legislation or common law concerning fraudulent acts or
- defrauding, attempting to defraud or conspiring to defraud the Client or
- any activity, practice or conduct which would constitute one of the offences listed above if such activity, practice or conduct had been carried out in the UK.

11.3 (19) Request for Information is a request for information or an apparent request under the Code of Practice on Access to government Information, FOIA or the Environmental Information Regulations.

11.3 (20) Relevant Requirements are all applicable Laws relating to bribery, corruption and fraud, including the Bribery Act 2010 and any guidance issued by the Secretary of State for Justice pursuant to section 9 of the Bribery Act 2010.

11.3 (21) Relevant Tax Authority is HM Revenue & Customs, or, if applicable, a tax authority in the jurisdiction in which the *Contractor* is established.

11.3 (22) Security Policy means the *Client's* security policy attached as Appendix 1 to Contract Schedule J (Security Provisions) as may be updated from time to time.

Option Z 4 - Admittance to site

Insert new clause 19A:

19A.1 The *Contractor* submits to the *Project Manager* details of people who are to be employed by it and its Subcontractors in Providing the Works. The details include a list of names and addresses, the capabilities in which they are employed, and other information required by the *Project Manager*.

19A.2 The *Project Manager* may instruct the *Contractor* to take measures to prevent unauthorised persons being admitted to the Site.

19A.3 Employees of the *Contractor* and its Subcontractors are to carry a *Client's* pass and comply with all conduct requirements from the *Client* whilst they are on the parts of the Site identified in the Scope.

19A.4 The *Contractor* submits to the *Project Manager* for acceptance a list of the names of the people for whom passes are required. On acceptance, the *Project Manager* issues the passes to the *Contractor*. Each pass is returned to the *Project Manager* when the person no longer requires access to that part of the Site or after the *Project Manager* has given notice that the person is not to be admitted to the Site.

19A.5 The *Contractor* does not take photographs of the Site or of work carried out in connection with the *works* unless it has obtained the acceptance of the *Project Manager*.

19A.6 The *Contractor* takes the measures needed to prevent its and its Subcontractors' people taking, publishing or otherwise circulating such photographs.

Option Z5 - Prevention of fraud and bribery

Insert new clauses:

18.4.1 The *Contractor* represents and warrants that neither it, nor to the best of its knowledge any of its people, have at any time prior to the Contract Date

- committed a Prohibited Act or been formally notified that it is subject to an investigation or prosecution which relates to an alleged Prohibited Act or
- been listed by any government department or agency as being debarred, suspended, proposed for suspension or debarment, or otherwise ineligible for participation in government procurement programmes or contracts on the grounds of a Prohibited Act.
 - 18.4.2 During the carrying out of the *works* the *Contractor* does not
- commit a Prohibited Act and

do or suffer anything to be done which would cause the *Client* or any of the *Client's* employees, consultants, contractors, sub-contractors or agents to contravene any of the Relevant Requirements or otherwise incur any liability in relation to the Relevant Requirements.

18.4.3 In Providing the Works the *Contractor*

- establishes, maintains and enforces, and requires that its Subcontractors establish, maintain and enforce, policies and procedures which are adequate to ensure compliance with the Relevant Requirements and prevent the occurrence of a Prohibited Act.
- keeps appropriate records of its compliance with this contract and make such records available to the *Client* on request and
- provides and maintains and where appropriate enforces an anti-bribery policy (which shall be disclosed to the *Client* on request) to prevent it and any *Contractor's* people or any person acting on the *Contractor's* behalf from committing a Prohibited Act.
 - 18.4.4 The *Contractor* immediately notifies the *Client* in writing if it becomes aware of any breach of clause 18.4.1, or has reason to believe that it has or any of its people or Subcontractors have
- been subject to an investigation or prosecution which relates to an alleged Prohibited Act.
- been listed by any government department or agency as being debarred, suspended, proposed for suspension or debarment, or otherwise ineligible for participation in government procurement programmes or contracts on the grounds of a Prohibited Act or
- received a request or demand for any undue financial or other advantage of any kind in connection with the performance of this contract or otherwise suspects that any person or party directly or indirectly connected with this contract has committed or attempted to commit a Prohibited Act.
 - 18.4.5 If the *Contractor* makes a notification to the *Client* pursuant to clause 18.4.4, the *Contractor* responds promptly to the *Client*'s enquiries, co-operates with any investigation, and allows the *Client* to audit any books, records and/or any other relevant documentation in accordance with this contract.
 - 18.4.6 If the *Contractor* breaches Clause 18.4.3, the *Client* may by notice require the *Contractor* to remove from carrying out the *works* any person whose acts or omissions have caused the *Contractor*'s breach.

Option Z7 - Legislation and Official secrets

Insert new clauses:

20.5 The Contractor complies with Law in the carrying out of the works.

20.6 The Official Secrets Acts 1911 to 1989 and, where appropriate, the provisions of section 11 of the Atomic Energy Act 1946 apply to this contract.

20.7 The *Contractor* notifies its employees and its Subcontractors of their duties under these Acts.

Option Z10 - Freedom of information

Insert new clauses:

29.3 The *Contractor* acknowledges that unless the *Project Manager* has notified the *Contractor* that the *Client* is exempt from the provisions of the FOIA, the *Client* is subject to the requirements of the Code of Practice on Government Information, the FOIA and the Environmental Information Regulations. The *Contractor* cooperates with and assists the *Client* so as to enable the *Client* to comply with its information disclosure obligations.

29.4 The Contractor

- transfers to the *Project Manager* all Requests for Information that it receives as soon
 as practicable and in any event within two working days of receiving a Request for
 Information,
- provides the *Project Manager* with a copy of all information in its possession, or power in the form that the *Project Manager* requires within five working days (or such other period as the *Project Manager* may specify) of the *Project Manager*'s request,
- provides all necessary assistance as reasonably requested by the *Project Manager* to enable the *Client* to respond to the Request for Information within the time for compliance set out in section 10 of the FOIA or regulation 5 of the Environmental Information Regulations and
- procures that its Subcontractors do likewise.
- 29.5 The *Client* is responsible for determining in its absolute discretion whether any information is exempt from disclosure in accordance with the provisions of the Code of Practice on Government Information, FOIA or the Environmental Information Regulations.
- 29.6 The *Contractor* does not respond directly to a Request for Information unless authorised to do so by the *Project Manager*.
- 29.7 The *Contractor* acknowledges that the *Client* may, acting in accordance with the Department of Constitutional Affairs' Code of Practice on the Discharge of the Functions of Public Authorities under Part 1 of the Freedom of information Act 2000, be obliged to disclose information without consulting or obtaining consent from the *Contractor* or despite the *Contractor* having expressed negative views when consulted.

29.8 The *Contractor* ensures that all information is retained for disclosure throughout the *period* for retention and permits the *Project Manager* to inspect such records as and when reasonably requested from time to time.

Option Z13 - Confidentiality and Information Sharing

Insert a new clause

29.9 Except to the extent set out in this clause or where disclosure is expressly permitted elsewhere in this contract, each Party shall

- treat the other Party's Confidential Information as confidential and safeguard it accordingly,
- not disclose the other Party's Confidential Information to any other person without prior written consent,
- immediately notify the other Party if it suspects unauthorised access, copying, use or disclosure of the Confidential Information and
- notify the Serious Fraud Office where the Party has reasonable grounds to believe that the other Party is involved in activity that may be a criminal offence under the Bribery Act 2010.

29.10 The clause above shall not apply to the extent that

- such disclosure is a requirement of the Law placed upon the party making the disclosure, including any requirements for disclosure under the FOIA or the Environmental Information Regulations pursuant to clause Z10 (Freedom of Information),
- such information was in the possession of the party making the disclosure without obligation of confidentiality prior to its disclosure by the information owner,
- such information was obtained from a third party without obligation of confidentiality,
- such information was already in the public domain at the time of disclosure otherwise than by a breach of this contract or
- it is independently developed without access to the other party's Confidential Information.
- 29.11 The *Contractor* may only disclose the *Client's* Confidential Information to the people who are directly involved in Providing the Works and who need to know the information, and shall ensure that such people are aware of and shall comply with these obligations as to confidentiality.

The *Contractor* shall not, and shall procure that the *Contractor's* people do not, use any of the Client Confidential Information received otherwise than for the purposes of this contract.

- 29.12 The Contractor may only disclose the Client Confidential Information to Contractor's people who need to know the information, and shall ensure that such people are aware of, acknowledge the importance of, and comply with these obligations as to confidentiality. In the event that any default, act or omission of any Contractor's people causes or contributes (or could cause or contribute) to the Contractor breaching its obligations as to confidentiality under or in connection with this contract, the *Contractor* shall take such action as may be appropriate in the circumstances, including the use of disciplinary procedures in serious cases. To the fullest extent permitted by its own obligations of confidentiality to any Contractor's people, the Contractor shall provide such evidence to the Client as the Client may reasonably require (though not so as to risk compromising or prejudicing the case) to demonstrate that the Contractor is taking appropriate steps to comply with this clause, including copies of any written communications to and/or from Contractor's people, and any minutes of meetings and any other records which provide an audit trail of any discussions or exchanges with Contractor's people in connection with obligations as to confidentiality.
- 29.13 At the written request of the *Client*, the *Contractor* shall procure that those members of the *Contractor's* people identified in the *Client's* request signs a confidentiality undertaking prior to commencing any work in accordance with this contract.

29.14 Nothing in this contract shall prevent the *Client* from disclosing the *Contractor*'s Confidential Information

- to any Crown Body or any other Contracting Bodies. All Crown Bodies or Contracting Bodies receiving such Confidential Information shall be entitled to further disclose the Confidential Information to other Crown Bodies or other Contracting Bodies on the basis that the information is confidential and is not to be disclosed to a third party which is not part of any Crown Body or any Contracting Body,
- to a professional adviser, contractor, consultant, supplier or other person engaged by the *Client* or any Crown Body (including any benchmarking organisation) for any purpose connected with this contract, or any person conducting an Office of Government Commerce Gateway Review,
- for the purpose of the examination and certification of the Client's accounts,
- for any examination pursuant to Section 6(1) of the National Audit Act 1983 of the economy, efficiency and effectiveness with which the *Client* has used its resources,
- for the purpose of the exercise of its rights under this contract or
- to a proposed successor body of the Client in connection with any assignment, novation or disposal of any of its rights, obligations or liabilities under this contract,

and for the purposes of the foregoing, disclosure of the Contractor's Confidential Information shall be on a confidential basis and subject to a confidentiality agreement or arrangement containing terms no less stringent than those placed on the *Client* under this clause 29.14.

- 29.15 The *Client* shall use all reasonable endeavours to ensure that any government department, Contracting Body, people, third party or subcontractor to whom the *Contractor's* Confidential Information is disclosed pursuant to the above clause is made aware of the *Client's* obligations of confidentiality.
- 29.16 Nothing in this clause shall prevent either party from using any techniques, ideas or know-how gained during the performance of the contract in the course of its normal business to the extent that this use does not result in a disclosure of the other party's Confidential Information or an infringement of IPR.
- 29.17 The Client may disclose the Confidential Information of the Contractor
 - to Parliament and Parliamentary Committees or if required by any Parliamentary reporting requirement,
 - to the extent that the *Client* (acting reasonably) deems disclosure necessary or appropriate in the course of carrying out its public functions,

Option Z14 - Security Requirements

The *Contractor* complies with, and procures the compliance of the *Contractor's* people, with the Security Policy and the Security Management Plan produced by the *Contractor* and the *Contractor* shall ensure that the Security Management Plan fully complies with the Security Policy and Contract Schedule J.

Option Z16 - Tax Compliance

Insert new clauses:

29.18 The *Contractor* represents and warrants that at the Contract Date, it has notified the *Client* in writing of any Occasions of Tax Non-Compliance or any litigation that it is involved in that is in connection with any Occasions of Tax Non-Compliance.

29.19 If, at any point prior to the *defects date*, an Occasion of Tax Non-Compliance occurs, the *Contractor* shall

- notify the Client in writing of such fact within 5 days of its occurrence and
- promptly provide to the Client

- details of the steps which the Contractor is taking to address the Occasions of Tax Non-Compliance and to prevent the same from recurring, together with any mitigating factors that it considers relevant and
- such other information in relation to the Occasion of Tax Non-Compliance as
 the Client may reasonably require.

Option Z22 - Fair payment

Insert a new clause:

57.1 The *Contractor* assesses the amount due to a Subcontractor without taking into account the amount certified by the *Project Manager*.

57.2 The Contractor includes in the contract with each Subcontractor

- a period for payment of the amount due to the Subcontractor not greater than 5 days after the final date for payment in this contract. The amount due includes, but is not limited to, payment for work which the Subcontractor has completed from the previous assessment date up to the current assessment date in this contract,
- a provision requiring the Subcontractor to include in each subsubcontract the same requirement (including this requirement to flow down, except that the period for payment is to be not greater than 9 days after the final date for payment in this contract and
- a provision requiring the Subcontractor to assess the amount due to a subsubcontractor without taking into account the amount paid by the *Contractor*.

Option Z42 - The Housing Grants, Construction and Regeneration Act 1996

Add an additional clause Y2.6

Y2.6

If Option Y(UK)2 is said to apply then notwithstanding that this contract relates to the carrying out of construction operations other than in England or Wales or Scotland, the Act is deemed to apply to this contract. [Guidance: for works carried out in Northern Ireland]

Option Z44 - Intellectual Property Rights Delete

clause 22 and insert the following clause

In this clause 22 only:

- "Document" means all designs, drawings, specifications, software, electronic data, photographs, plans, surveys, reports, and all other documents and/or information prepared by or on behalf of the *Contractor* in relation to this contract.
- 22.1 The Intellectual Property Rights in all Documents prepared by or on behalf of the *Contractor* in relation to this contract and the work executed from them remains the property of the *Contractor*. The *Contractor* hereby grants to the *Client* an irrevocable, royalty free, non-exclusive licence to use and reproduce the Documents for any and all purposes connected with the construction, use, alterations or demolition of the *works*. Such licence entitles the *Client* to grant sub-licences to third parties in the same terms as this licence provided always that the *Contractor* shall not be liable to any licencee for any use of the Documents or the Intellectual Property Rights in the Documents for purposes other than those for which the same were originally prepared by or on behalf of the *Contractor*.
- 22.2 The *Client* may assign novate or otherwise transfer its rights and obligations under the licence granted pursuant to 22.1 to a Crown Body or to anybody (including any private sector body) which performs or carries on any functions and/or activities that previously had been performed and/or carried on by the *Client*.
- 22.3 In the event that the *Contractor* does not own the copyright or any Intellectual Property Rights in any Document the *Contractor* uses all reasonable endeavours to procure the right to grant such rights to the *Client* to use any such copyright or Intellectual Property Rights from any third party owner of the copyright or Intellectual Property Rights. In the event that the *Contractor* is unable to procure the right to grant to the *Client* in accordance with the foregoing the *Contractor* procures that the third party grants a direct licence to the *Client* on industry acceptable terms.
- 22.4 The *Contractor* waives any moral right to be identified as author of the Documents in accordance with section 77, Copyright Designs and Patents Acts 1988 and any right not to have the Documents subjected to derogatory treatment in accordance with section 8 of that Act as against the *Client* or any licensee or assignee of the *Client*.
- 22.5 In the event that any act unauthorised by the *Client* infringes a moral right of the *Contractor* in relation to the Documents the *Contractor* undertakes, if the *Client* so requests and at the *Client*'s expense, to institute proceedings for infringement of the moral rights.
- 22.6 The *Contractor* warrants to the *Client* that it has not granted and shall not (unless authorised by the *Client*) grant any rights to any third party to use or otherwise exploit the Documents.
- 22.7 The *Contractor* supplies copies of the Documents to the *Project Manager* and to the *Client*'s other contractors and consultants for no additional fee to the extent necessary to enable them to discharge their respective functions in relation to this contract or related works.
- 22.8 After the termination or conclusion of the *Contractor*'s employment hereunder, the *Contractor* supplies the *Project Manager* with copies and/or computer discs of such of the

Documents as the *Project Manager* may from time to time request and the *Client* pays the *Contractor*'s reasonable costs for producing such copies or discs.

22.9 In carrying out the works the Contractor does not infringe any Intellectual Property

Rights of any third party. The *Contractor* indemnifies the *Client* against claims, proceedings, compensation and costs arising from an infringement or alleged infringement of the Intellectual Property Rights of any third party.

Option Z46 - MoD DEFCON Requirements

Insert a new clause:

19B

This clause is to incorporate MoD special terms and conditions in the form of DEFCONs and DEFORMs as detailed in [Guidance: Client to reference DEFCON / DEFORM Schedule].

Option Z47 - Small and Medium Sized Enterprises (SMEs)

Insert new clause:

26.5

The Contractor is required to take all reasonable steps to engage SMEs as Subcontractors and to seek to ensure that no less than the SME percentage of Subcontractors stated in the Contract Data are SMEs or that a similar proportion of the Defined Cost is undertaken by SMEs.

The *Contractor* is required to report to the *Client* in its regular contract management monthly reporting cycle the numbers of SMEs engaged as Subcontractors and the value of the Defined Cost that has been undertaken by SMEs.

Where available, the *Contractor* is required to tender its Subcontracts using the same online electronic portal as was provided by the *Client* for the purposes of tendering this contract.

The *Contractor* is to ensure that the terms and conditions used to engage Subcontractors are no less favourable than those of this contract. A reason for the *Project Manager* not accepting subcontract documents proposed by the *Contractor* is that they are unduly disadvantageous to the Subcontractor.

Option Z48 - Apprenticeships

Insert new clause:

26.6

The *Contractor* takes all reasonable steps to employ apprentices, and reports to the *Client* the numbers of apprentices employed and the wider skills training provided, during the delivery of the *works*.

The *Contractor* takes all reasonable steps to ensure that no less than a percentage of its people (agreed between the Parties) are on formal apprenticeship programmes or that a similar proportion of hours worked in Providing the Works, (which may include support staff and Subcontractors) are provided by people on formal apprenticeship programmes.

The *Contractor* makes available to its people and Subcontractors working on the contract, information about the Government's Apprenticeship programme and wider skills opportunities.

The *Contractor* provides any further skills training opportunities that are appropriate for its people engaged in Providing the Works.

The *Contractor* provides a report detailing the following measures in its regular contract management monthly reporting cycle and is prepared to discuss apprenticeships at its regular meetings with the *Project Manager*

- the number of people during the reporting period employed on the contract, including support staff and Subcontractors,
- the number of apprentices and number of new starts on apprenticeships directly initiated through this

contract,

- the percentage of all people taking part in an apprenticeship programme,
- if applicable, an explanation from the *Contractor* as to why it is not managing to meet the specified percentage target,
- actions being taken to improve the take up of apprenticeships and
- other training/skills development being undertaken by people in relation to this contract, including:
 - (a) work experience placements for 14 to 16 year olds,
 - (b) work experience /work trial placements for other ages,
 - (c) student sandwich/gap year placements,

- (d) graduate placements,
- (e) vocational training,
- (f) basic skills training and
- (g) on site training provision/ facilities.

Option Z100 - GDPR

Insert new clause Z100 as follows: Z100

GDPR

The *Client* and the *Contractor* shall comply with the provisions of schedule [Guidance: insert schedule ref here]

Option Z101 – Cyber Essentials

Insert new clause Z101 as follows:

Z101 Cyber Essentials

The *Client* and the *Contractor* shall comply with the provisions of schedule [Guidance: insert schedule ref here]

SCHEDULE[Guidance: insert schedule ref here]

GDPR

The following definitions shall apply to this Schedule [Guidance: insert schedule ref here]

Agreement: this contract;

Processor Personnel: means all directors, officers, employees, agents, consultants and contractors of the Processor and/or of any Sub-Processor engaged in the performance of its obligations under this Agreement

GDPR CLAUSE DEFINITIONS:

Data Protection Legislation: (i) the GDPR, the LED and any applicable national implementing Laws as amended from time to time (ii) the DPA 2018 subject to Royal Assent to the extent that it relates to processing of personal data and privacy; (iii) all applicable Law about the processing of personal data and privacy;

Data Protection Impact Assessment : an assessment by the Controller of the impact of the envisaged processing on the protection of Personal Data.

Controller, Processor, Data Subject, Personal Data, Personal Data Breach, Data Protection Officer take the meaning given in the GDPR.

Data Loss Event: any event that results, or may result, in unauthorised access to Personal Data held by the Processor under this Agreement, and/or actual or potential loss and/or destruction of Personal Data in breach of this Agreement, including any Personal Data Breach.

Data Subject Request: a request made by, or on behalf of, a Data Subject in accordance with rights granted pursuant to the Data Protection Legislation to access their Personal Data.

DPA 2018: Data Protection Act 2018

GDPR: the General Data Protection Regulation (Regulation (EU) 2016/679)

Joint Controllers: where two or more Controllers jointly determine the purposes and means of processing

LED: Law Enforcement Directive (Directive (EU) 2016/680)

Protective Measures: appropriate technical and organisational measures which may include: pseudonymising and encrypting Personal Data, ensuring confidentiality, integrity, availability and resilience of systems and services, ensuring that availability of and access to Personal Data can be restored in a timely manner after an incident, and regularly assessing and evaluating the effectiveness of the such measures adopted by it including those outlined in Schedule [x] (Security).

Sub-processor: any third party appointed to process Personal Data on behalf of that Processor related to this Agreement

1. DATA PROTECTION

- 1.1 The Parties acknowledge that for the purposes of the Data Protection Legislation, the *Client* is the Controller and the *Contractor* is the Processor unless otherwise specified in Schedule
- [X]. The only processing that the Processor is authorised to do is listed in Schedule [X] by the Controller and may not be determined by the Processor.
- 1.2 The Processor shall notify the Controller immediately if it considers that any of the Controller's instructions infringe the Data Protection Legislation.
- 1.3 The Processor shall provide all reasonable assistance to the Controller in the preparation of any Data Protection Impact Assessment prior to commencing any processing. Such assistance may, at the discretion of the Controller, include:

- (a) a systematic description of the envisaged processing operations and the purpose of the processing;
- (b) an assessment of the necessity and proportionality of the processing operations in relation to the *works*;
- (c) an assessment of the risks to the rights and freedoms of Data Subjects; and
- (d) the measures envisaged to address the risks, including safeguards, security measures and mechanisms to ensure the protection of Personal Data.
- 1.4 The Processor shall, in relation to any Personal Data processed in connection with its obligations under this Agreement:
- (a) process that Personal Data only in accordance with Schedule [X], unless the Processor is required to do otherwise by Law. If it is so required the Processor shall promptly notify the Controller before processing the Personal Data unless prohibited by Law;
- (b) ensure that it has in place Protective Measures, are appropriate to protect against a Data Loss Event, which the Controller may reasonably reject (but failure to reject shall not amount to approval by the Controller of the adequacy of the Protective Measures), having taken account of the:
 - (i) nature of the data to be protected;
 - (ii) harm that might result from a Data Loss Event;
 - (iii) state of technological development; and
 - (iv) cost of implementing any measures;
- (c) ensure that:
- (i) the Processor Personnel do not process Personal Data except in accordance with this Agreement (and in particular Schedule X);
- (ii) it takes all reasonable steps to ensure the reliability and integrity of any Processor Personnel who have access to the Personal Data and ensure that they:
 - (A) are aware of and comply with the Processor's duties under this clause;
 - (B) are subject to appropriate confidentiality undertakings with the Processor or any Subprocessor;
 - (C) are informed of the confidential nature of the Personal Data and do not publish, disclose or divulge any of the Personal Data to any third Party unless directed in writing to do so by the Controller or as otherwise permitted by this Agreement; and
 - (D) have undergone adequate training in the use, care, protection and handling of Personal Data; and
 - (d) not transfer Personal Data outside of the EU unless the prior written consent of the Controller has been obtained and the following conditions are fulfilled:

- the Controller or the Processor has provided appropriate safeguards in relation to the transfer (whether in accordance with GDPR Article 46 or LED Article 37) as determined by the Controller;
- (ii) the Data Subject has enforceable rights and effective legal remedies;
- (iii) the Processor complies with its obligations under the Data Protection Legislation by providing an adequate level of protection to any Personal Data that is transferred (or, if it is not so bound, uses its best endeavours to assist the Controller in meeting its obligations); and
- (iv) the Processor complies with any reasonable instructions notified to it in advance by the Controller with respect to the processing of the Personal Data;
- (e) at the written direction of the Controller, delete or return Personal Data (and any copies of it) to the Controller on termination of the Agreement unless the Processor is required by Law to retain the Personal Data.
- 1.5 Subject to clause 1.6, the Processor shall notify the Controller immediately if it:
- (a) receives a Data Subject Request (or purported Data Subject Request);
- (b) receives a request to rectify, block or erase any Personal Data;
- (c) receives any other request, complaint or communication relating to either Party's obligations under the Data Protection Legislation;
- (d) receives any communication from the Information Commissioner or any other regulatory authority in connection with Personal Data processed under this Agreement;
- (e) receives a request from any third Party for disclosure of Personal Data where compliance with such request is required or purported to be required by Law; or
- (f) becomes aware of a Data Loss Event.
- 1.6 The Processor's obligation to notify under clause 1.5 shall include the provision of further information to the Controller in phases, as details become available.
- 1.7 Taking into account the nature of the processing, the Processor shall provide the

Controller with full assistance in relation to either Party's obligations under Data Protection Legislation and any complaint, communication or request made under clause 1.5 (and insofar as possible within the timescales reasonably required by the Controller) including by promptly providing:

- (a) the Controller with full details and copies of the complaint, communication or request;
- (b) such assistance as is reasonably requested by the Controller to enable the Controller to comply with a Data Subject Request within the relevant timescales set out in the Data Protection Legislation;
- (c) the Controller, at its request, with any Personal Data it holds in relation to a Data Subject;
- (d) assistance as requested by the Controller following any Data Loss Event;

- (e) assistance as requested by the Controller with respect to any request from the Information Commissioner's Office, or any consultation by the Controller with the Information Commissioner's Office.
 - 1.8 The Processor shall maintain complete and accurate records and information to demonstrate its compliance with this clause. This requirement does not apply where the Processor employs fewer than 250 staff, unless:
 - (a) the Controller determines that the processing is not occasional;
 - (b) the Controller determines the processing includes special categories of data as referred to in Article 9(1) of the GDPR or Personal Data relating to criminal convictions and offences referred to in Article 10 of the GDPR; or
 - (c) the Controller determines that the processing is likely to result in a risk to the rights and freedoms of Data Subjects.
- 1.9 The Processor shall allow for audits of its Data Processing activity by the Controller or the Controller's designated auditor.
- 1.10 Each Party shall designate its own data protection officer if required by the Data Protection Legislation .
- 1.11 Before allowing any Sub-processor to process any Personal Data related to this Agreement, the Processor must:
 - (a) notify the Controller in writing of the intended Sub-processor and processing;
 - (b) obtain the written consent of the Controller:
 - (c) enter into a written agreement with the Sub-processor which give effect to the terms set out in this clause [X] such that they apply to the Sub-processor; and
 - (d) provide the Controller with such information regarding the Sub-processor as the Controller may reasonably require.
 - 1.12 The Processor shall remain fully liable for all acts or omissions of any of its Subprocessors.
 - 1.13 The Controller may, at any time on not less than 30 Working Days' notice, revise this clause by replacing it with any applicable controller to processor standard clauses or similar terms forming part of an applicable certification scheme (which shall apply when incorporated by attachment to this Agreement).
 - 1.14 The Parties agree to take account of any guidance issued by the Information

Commissioner's Office. The Controller may on not less than 30 Working Days' notice to the Processor amend this agreement to ensure that it complies with any guidance issued by the Information Commissioner's Office.

1.15 Where the Parties include two or more Joint Controllers as identified in Schedule [X] in accordance with GDPR Article 26, those Parties shall enter into a Joint Controller

Agreement based on the terms outlined in Schedule [Y] in replacement of Clauses 1.1-1.14 for the Personal Data under Joint Control.

Annex A - Part 2: Schedule of Processing, Personal Data and Data Subjects

Schedule [X] Processing, Personal Data and Data Subjects

This Schedule shall be completed by the Controller, who may take account of the view of the Processors, however the final decision as to the content of this Schedule shall be with the Controller at its absolute discretion.

- 1. The contact details of the Controller's Data Protection Officer are: [Insert Contact details]
- 2. The contact details of the Processor's Data Protection Officer are: [Insert Contact details]
- 3. The Processor shall comply with any further written instructions with respect to processing by the Controller.
- 4. Any such further instructions shall be incorporated into this Schedule.

Description	Details
Identity of the Controller and Processor	The Parties acknowledge that for the purposes of the Data Protection Legislation, the <i>Client</i> is the Controller and the <i>Contractor</i> is the Processor in accordance with Clause 1.1. [Guidance: You may need to vary this section where (in the rare case) the <i>Client</i> and <i>Contractor</i> have a different relationship. For example where the Parties are Joint Controller of some Personal Data: "Notwithstanding Clause 1.1 the Parties acknowledge that they are also Joint Controllers for the purposes of the Data Protection Legislation in respect of: [Insert the scope of Personal Data which the purposes and means of the processing is determined by the both Parties] In respect of Personal Data under Joint Control, Clause 1.1-1.15 will not apply and the Parties agree to put in place a Joint Controller Agreement as outlined in Schedule Y instead."

Subject matter of the processing	[This should be a high level, short description of what the processing is about i.e. its subject matter of the contract. Example: The processing is needed in order to ensure that the Processor can effectively deliver the contract to provide a service to
	members of the public.]
Duration of the processing	[Clearly set out the duration of the processing including dates]
Nature and purposes of the processing	[Please be as specific as possible, but make sure that you cover all intended purposes.
	The nature of the processing means any operation such as collection, recording, organisation, structuring, storage, adaptation or alteration, retrieval, consultation, use, disclosure by transmission, dissemination or otherwise making available, alignment or combination, restriction, erasure or destruction of data (whether or not by automated means) etc.
	The purpose might include: employment processing, statutory obligation, recruitment assessment etc]
Type of Personal Data being Processed	[Examples here include: name, address, date of birth, NI number, telephone number, pay, images, biometric data etc]
Categories of Data Subject	
	[Examples include: Staff (including volunteers, agents, and temporary workers), customers/ clients, suppliers, patients, students / pupils, members of the public, users of a particular website etc]

[Describe how long the data will be retained for, how it be returned or destroyed]

Plan for return and destruction of the data once the processing is complete

UNLESS requirement

SCHEDULE J

1. CONTRACT SCHEDULE J - SECURITY PROVISIONS

1.1 Definitions

type of data

under union or member state law to preserve that

For the purposes of this schedule the following terms shall have the meanings given below:

"Affiliates"

in relation to a body corporate, any other entity which directly or indirectly Controls, is Controlled by, or is under direct or indirect common Control with, that body corporate from time to time;

"Breach of Security"

in accordance with the Security Requirements and the Security Policy, the occurrence of:

- (a) any unauthorised access to or use of the works the Client Premises, the Sites, the Contractor System and/or any ICT, information or data (including the Confidential Information and the Client Data) used by the *Client* and/or the *Contractor* in connection with this contract; and/or
- (b) the loss and/or unauthorised disclosure of any information or data (including the Confidential Information and the Client Data), including any copies of such information or data, used by the *Client* and/or the *Contractor* in connection with this contract.

"Clearance"

means national security clearance and employment checks undertaken by and/or obtained from the Defence Vetting Agency;

"Contractor Equipment"

the hardware, computer and telecoms devices and equipment supplied by the *Contractor* or its Subcontractors (but not hired, leased or loaned from the *Client*) for the carrying out of the *works*;

"Contractor Software"

software which is proprietary to the *Contractor*, including software which is or will be used by the *Contractor* for the purposes of carrying out of the *works*;

"Contractor System"

the information and communications technology system used by the *Contractor* in carrying out of the *works* including the Software, the *Contractor* Equipment and related cabling (but excluding the Client System);

"Control"

means that a person possesses, directly or indirectly, the power to direct or cause the direction of the management

and policies of the other person (whether through the ownership of voting shares, by contract or otherwise) and

"Default"

"Controls" and "Controlled" shall be interpreted accordingly;

any breach of the obligations of the relevant party (including but not limited to fundamental breach or breach of a fundamental term) or any other default, act, omission, negligence or statement of the relevant party, its employees, servants, agents or Sub contractors in connection with or in relation to the subject-matter of this contract and in respect of which such party is liable to the other;

"Dispute Resolution Procedure"

the dispute resolution procedure set out in this contract (if any) or as agreed between the parties;

"Client Premises"

means premises owned, controlled or occupied by the *Client* or its Affiliates which are made available for use by the *Contractor* or its Subcontractors for carrying out of the *works* (or any of them) on the terms set out in this contract or any separate agreement or licence;

"Client System"

the *Client*'s computing environment (consisting of hardware, software and/or telecommunications networks or equipment) used by the *Client* or the *Contractor* in connection with this contract which is owned by or licensed to the *Client* by a third party and which interfaces with the *Contractor* System or which is necessary for the *Client* to receive the *works*;

"Environmental Information Regulations"

the Environmental Information Regulations 2004 together with any guidance and/or codes of practice issues by the Information Commissioner or relevant Government

Department in relation to such regulations;

"FOIA"

the Freedom of Information Act 2000 and any subordinate legislation made under this Act from time to time together with any guidance and/or codes of practice issued by the Information Commissioner or relevant Government

Department in relation to such legislation;

"Good Industry

Practice" the exercise of that degree of skill, care, prudence, efficiency,

foresight and timeliness as would be expected from a leading

company within the relevant industry or business sector;

"ICT" information and communications technology;

"ICT Environment" the Client System and the Contractor System;

"Impact Assessment" an assessment of a Compensation Event;

"Information" has the meaning given under section 84 of the Freedom of

Information Act 2000;

"Information Assets

Register"

the register of information assets to be created and maintained by the *Contractor* throughout the carrying out of the *works* as described in the contract (if any) or as otherwise agreed between

the parties;

"ISMS" the Information Security Management System as defined by

ISO/IEC 27001. The scope of the ISMS will be as agreed by the

parties and will directly reflect the scope of the works;

"Know-How" all ideas, concepts, schemes, information, knowledge,

techniques, methodology, and anything else in the nature of know how relating to the *works* but excluding know how already in the *Contractor's* or the *Client's* possession before this

contract;

"List x"

means, in relation to a Subcontractor, one who has been placed on List x in accordance with Ministry of Defence guidelines and procedures, due to that Sub contractor undertaking work on its premises marked as CONFIDENTIAL or above;

"Malicious Software"

any software program or code intended to destroy, interfere with, corrupt, or cause undesired effects on program files, data or other information, executable code or application software macros, whether or not its operation is immediate or delayed, and whether the malicious software is introduced wilfully, negligently or without knowledge of its existence;

"Process" has the meaning given to it under the Data Protection Legislation

but, for the purposes of this contract, it shall include

both manual and automatic processing;

"Protectively Marked" shall have the meaning as set out in the Security Policy

Framework.

"Regulatory Bodies"

those government departments and regulatory, statutory and other entities, committees and bodies which, whether under statute, rules, regulations, codes of practice or otherwise, are entitled to regulate, investigate, or influence the matters dealt with in this contract or any other affairs of the Client and "Regulatory Body" shall be construed accordingly;

"Request for Information" a request for information or an apparent request under the Code of Practice on Access to Government Information, FOIA or the **Environmental Information Regulations**;

"Security Management Plan"

the Contractor's security plan prepared pursuant to paragraph 1.5.3 of schedule J (Security Management Plan) an outline of which is set out in Appendix 1 of schedule J (Security Management Plan);

"Security Policy Framework"

means the Cabinet Office Security Policy Framework (available from the Cabinet Office Security Policy Division);

"Security Requirements"

means the requirements in the contract relating to security of the carrying out of the works (if any) or such other requirements as the *Client* may notify to the *Contractor* from time to time

"Security Tests"

shall have the meaning set out in Appendix 2 (Security

Management Plan) [Guidance: define "Security Tests" in Security Management Plan]

"Software" Specially Written Software, Contractor Software and Third Party

Software;

"Specially Written

Software"

any software created by the Contractor (or by a third party on behalf of the Contractor) specifically for the purposes of this

contract;

"Staff Vetting **Procedures**"

the Client's procedures and departmental policies for the vetting of personnel whose role will involve the handling of information of a sensitive or confidential nature or the handling of information which is subject to any relevant security measures, including, but not limited to, the provisions of the Official Secrets Act 1911 to 1989;

"Statement of Applicability" shall have the meaning set out in ISO/IEC 27001 and as agreed by the parties during the procurement phase; "Standards"

> the British or international standards, Client's internal policies and procedures, Government codes of practice and guidance together with any other specified policies or procedures referred to in this contract (if any) or as otherwise agreed by the parties;

"Third Party Software"

software which is proprietary to any third party other than an Affiliate of the Contractor which is or will be used by the Contractor for the purposes of carrying out of the works; and

1.2 Introduction

- 1.2.1 This schedule covers:
 - 1.2.1.1 principles of protective security to be applied in carrying out of the works;
 - 1.2.1.2 wider aspects of security relating to carrying out of the works;
 - 1.2.1.3 the development, implementation, operation, maintenance and continual improvement of an ISMS;
 - 1.2.1.4 the creation and maintenance of the Security Management Plan:
 - 1.2.1.5 audit and testing of ISMS compliance with the Security Requirements;
 - 1.2.1.6 conformance to ISO/IEC 27001
 (Information Security Requirements Specification) and ISO/IEC27002 (Information Security Code of Practice) and;
 - 1.2.1.7 obligations in the event of actual, potential or attempted breaches of security.

1.3 Principles of Security

- 1.3.1 The *Contractor* acknowledges that the *Client* places great emphasis on the confidentiality, integrity and availability of information and consequently on the security provided by the ISMS.
- 1.3.2 The *Contractor* shall be responsible for the effective performance of the ISMS and shall at all times provide a level of security which:
 - 1.3.2.1 is in accordance with Good Industry Practice, the *law of the contract* and this contract;
 - 1.3.2.2 complies with the Security Policy;
 - 1.3.2.3 complies with at least the minimum set of security measures and standards as determined by the Security Policy Framework (Tiers 1-4) available from the Cabinet Office Security Policy Division (COSPD);
 - 1.3.2.4 meets any specific security threats to the ISMS; and
 - 1.3.2.5 complies with ISO/IEC27001 and ISO/IEC27002 in accordance with paragraph 1.3.2 of this schedule; 1.3.2.6 complies with the Security Requirements; and
 - 1.3.2.7 complies with the *Client's* ICT standards.

- 1.3.3 The references to standards, guidance and policies set out in paragraph
 1.3.2.2 shall be deemed to be references to such items as developed and
 updated and to any successor to or replacement for such standards,
 guidance and policies, from time to time.
- 1.3.4 In the event of any inconsistency in the provisions of the above standards, guidance and policies, the *Contractor* gives an early warning to the *Project Manager* of such inconsistency immediately upon becoming aware of the same, and the *Project Manager* shall, as soon as practicable, advise the *Contractor* which provision the *Contractor* shall be required to comply with.

1.4 ISMS and Security Management Plan

1.4.1 Introduction:

- (i) The *Contractor* shall develop, implement, operate, maintain and continuously improve and maintain an ISMS which will, without prejudice to paragraph 1.3.2, be accepted, by the *Project Manager*, tested in accordance with the provisions relating to testing as set out in the contract (if any) or as
 - otherwise agreed between the Parties, periodically updated and audited in accordance with ISO/IEC 27001.
 - 1.4.1.1 The *Contractor* shall develop and maintain a Security Management Plan in accordance with this Schedule to apply during the carrying out of the *works*.
 - 1.4.1.2 The *Contractor* shall comply with its obligations set out in the Security Management Plan.
 - 1.4.1.3 Both the ISMS and the Security Management Plan shall, unless otherwise specified by the *Client*, aim to protect all aspects of the *works* and all processes associated with carrying out of the *works*, including the construction, use, alterations or demolition of the *works*, the *Contractor* System and any ICT, information and data (including the Client Confidential Information and the Client Data) to the extent used by the *Client* or the *Contractor* in connection with this contract.

1.4.2 Development of the Security Management Plan:

1.4.2.1 Within 20 Working Days after the Contract Date and in accordance with paragraph 1.4.4 (Amendment and Revision), the Contractor will prepare and deliver to the Project Manager for acceptance a fully complete and up to date Security Management Plan which will be based on the draft Security

Management Plan set out in Appendix 2 of this Part 2 of this Contract Schedule J.

- 1.4.2.2 If the Security Management Plan, or any subsequent revision to it in accordance with paragraph 1.4.4 (Amendment and Revision), is accepted by the Project Manager it will be adopted immediately and will replace the previous version of the Security Management Plan at Appendix 2 of this Part 2 of this Contract Schedule J. If the Security Management Plan is not accepted by the Project Manager the Contractor shall amend it within 10 Working Days or such other period as the parties may agree in writing of a notice of non-acceptance from the Project Manager and re-submit to the Project Manager for accepted. The parties will use all reasonable endeavours to ensure that the acceptance process takes as little time as possible and in any event no longer than 15 Working Days (or such other period as the parties may agree in writing) from the date of its first submission to the *Project* Manager. If the Project Manager does not accept the Security Management Plan following its resubmission, the matter will be resolved in accordance with the Dispute Resolution Procedure. No acceptance to be given by the Project *Manager* pursuant to this paragraph 1.4.2.2 of this schedule may be unreasonably withheld or delayed. However any failure to accept the Security Management Plan on the grounds that it does not comply with the requirements set out in paragraph 1.4.3.4 shall be deemed to be reasonable.
- 1.4.3 Content of the Security Management Plan:
 - 1.4.3.1 The Security Management Plan will set out the security measures to be implemented and maintained by the *Contractor* in relation to all aspects of the *works* and all processes associated with carrying out of the *works* and shall at all times comply with and specify security measures and procedures which are sufficient to ensure that the *works* comply with the provisions of this schedule (including the principles set out in paragraph 1.3);
 - 1.4.3.2 The Security Management Plan (including the draft version) should also set out the plans for transiting all security arrangements and responsibilities from those in place at the Contract Date to those incorporated in the Contractor's ISMS at the date notified by the Project Manager to the Contractor for the Contractor to meet the full obligations of the Security Requirements.
 - 1.4.3.3 The Security Management Plan will be structured in accordance with ISO/IEC27001 and ISO/IEC27002, crossreferencing if necessary to other schedules of this contract which cover specific areas included within that standard.

1.4.3.4 The Security Management Plan shall be written in plain English in language which is readily comprehensible to the staff of the

Contractor and the Client engaged in the works and shall only reference documents which are in the possession of the Client or whose location is otherwise specified in this schedule.

- 1.4.4 Amendment and Revision of the ISMS and Security Management Plan:
 - 1.4.4.1 The ISMS and Security Management Plan will be fully reviewed and updated by the *Contractor* annually or from time to time to reflect:
 - (a) emerging changes in Good Industry Practice;
 - (b) any change or proposed change to the ContractorSystem, the works and/or associated processes;
 - (c) any new perceived or changed security threats; and
 - (d) any reasonable request by the *Project Manager*.
 - 1.4.4.2 The *Contractor* will provide the *Project Manager* with the results of such reviews as soon as reasonably practicable after their completion and amend the ISMS and Security Management Plan at no additional cost to the *Client*. The results of the review should include, without limitation:
 - (a) suggested improvements to the effectiveness of the ISMS;
 - (b) updates to the risk assessments;
 - (c) proposed modifications to the procedures and controls that effect information security to respond to events that may impact on the ISMS; and
 - (d) suggested improvements in measuring the effectiveness of controls.
 - 1.4.4.3 On receipt of the results of such reviews, the *Project Manager* will accept any amendments or revisions to the ISMS or Security Management Plan in accordance with the process set out at paragraph 1.4.2.2.
 - 1.4.4.4 Any change or amendment which the Contractor proposes to make to the ISMS or Security Management Plan (as a result of a Project Manager's request or change to the works or otherwise) shall be subject to the early warning procedure and

shall not be implemented until accepted in writing by the *Project Manager*.

1.4.5 Testing

- 1.4.5.1 The Contractor shall conduct Security Tests of the ISMS on an annual basis or as otherwise agreed by the parties. The date, timing, content and conduct of such Security Tests shall be agreed in advance with the Project Manager.
- 1.4.5.2 The *Project Manager* shall be entitled to witness the conduct of the Security Tests. The *Contractor* shall provide the *Project Manager* with the results of such tests (in a form accepted by the *Client* in advance) as soon as practicable after completion of each Security Test.
- 1.4.5.3 Without prejudice to any other right of audit or access granted to the Client pursuant to this contract, the Project Manager and/or its authorised representatives shall be entitled, at any time and without giving notice to the Contractor, to carry out such tests (including penetration tests) as it may deem necessary in relation to the ISMS and the Contractor's compliance with the ISMS and the Security Management Plan. The Project Manager may notify the Contractor of the results of such tests after completion of each such test. Security Tests shall be designed and implemented so as to minimise the impact on the carrying out of the works. If such tests adversely affect the Contractor's ability to carry out the works in accordance with the Scope, the Contractor shall be

granted relief against any resultant under-performance for the period of the tests.

1.4.5.4 Where any Security Test carried out pursuant to paragraphs 1.4.5.2 or 1.4.5.3 above reveals any actual or potential Breach of Security, the *Contractor* shall promptly notify the *Project Manager* of any changes to the ISMS and to the Security Management Plan (and the implementation thereof) which the *Contractor* proposes to make in order to correct such failure or weakness. Subject to the *Project Manager's* acceptance in accordance with paragraph (i), the *Contractor* shall implement such changes to the ISMS and the Security Management Plan in accordance with the timetable agreed with the *Project Manager* or, otherwise, as soon as reasonably possible. Where the change to the ISMS or Security Management Plan is made to address a noncompliance with the Security Policy or Security Requirements, the change to the ISMS or Security Management Plan is Disallowed Cost.

1.5 Compliance with ISO/IEC 27001

1.5.1 Unless otherwise agreed by the parties, the *Contractor* shall obtain independent certification of the ISMS to ISO/IEC 27001 within 12 months of

the Contract Date and shall maintain such certification until the Defects Certificate or a termination certificate has been issued.

1.5.2 In the event that paragraph 1.5.1 above applies, if certain parts of the ISMS do not conform to Good Industry Practice, or controls as described in

ISO/IEC 27002 are not consistent with the Security Policy, and, as a result, the *Contractor* reasonably believes that it is not compliant with ISO/IEC 27001, the *Contractor* shall promptly notify the *Project Manager* of this and the *Client* in its absolute discretion may waive the requirement for certification in respect of the relevant parts.

- 1.5.3 The *Project Manager* shall be entitled to carry out such regular security audits as may be required and in accordance with Good Industry Practice, in order to ensure that the ISMS maintains compliance with the principles and practices of ISO 27001.
- 1.5.4 If, on the basis of evidence provided by such audits, it is the *Project Manager*'s reasonable opinion that compliance with the principles and practices of ISO/IEC 27001 is not being achieved by the *Contractor*, then the *Project Manager* shall notify the *Contractor* of the same and give the *Contractor* a reasonable time (having regard to the extent and criticality of any non-compliance and any other relevant circumstances) to become compliant with the principles and practices of ISO/IEC 27001. If the *Contractor* does not become compliant within the required time then the *Project Manager* has the right to obtain an independent audit against these standards in whole or in part.
- 1.5.5 If, as a result of any such independent audit as described in paragraph 1.5.4 the *Contractor* is found to be non-compliant with the principles and practices of ISO/IEC 27001 then the *Contractor* shall, at its own expense, undertake those actions required in order to achieve the necessary compliance and shall reimburse in full the costs incurred by the *Client* in obtaining such audit.

1.6 Breach of Security

- 1.6.1 Either party shall give an early warning to the other in accordance with the agreed security incident management process as defined by the ISMS upon becoming aware of any Breach of Security or any potential or attempted Breach of Security.
- 1.6.2 Without prejudice to the security incident management process, upon becoming aware of any of the circumstances referred to in paragraph 1.6.1, the *Contractor* shall:
 - 1.6.2.1 immediately take all reasonable steps necessary to:
 - remedy such breach or protect the integrity of the ISMS against any such potential or attempted breach or threat; and

(b) prevent an equivalent breach in the future.

such steps shall include any action or changes reasonably required by the *Project Manager*, and

1.6.2.2 as soon as reasonably practicable provide to the *Project Manager* full details (using such reporting mechanism as defined by the ISMS) of the Breach of Security or the potential or attempted Breach of Security.

Appendix 1 – Security Policy

[Guidance Note: Append Security Policy]

Appendix 2 - Security Management Plan

[Guidance Note: Append Security Management Plan] SCHEDULE [Guidance: insert schedule ref here] CYBER ESSENTIALS

CYBER ESSENTIALS SCHEME

1. DEFINITIONS

1.1 In this Schedule, the following words shall have the following meanings:

"Cyber Essentials Scheme"

Certificate"

the Cyber Essentials Scheme developed by the

Government which provides a clear statement of the basic controls all organisations should implement to mitigate the risk from common internet based threats (as may be amended from time to time). Details

of the Cyber Essentials

Scheme can be found

here:https://www.gov.uk/governm

ent/

publications/cyberessentialsscheme-

overview:

"Cyber Essentials Basic the certificate awarded on the basis of

Self-assess

self-assessment, verified by an

independent

certification body, under the Cyber Essentials Scheme and is the basic

level of assurance;

"Cyber Essentials Certificate"

Cyber Essentials Basic Certificate, the Cyber Essentials Plus Certificate or the Cyber Essential Scheme certificate equivalent to be provided by the Contractor as set out in the Framework Data Sheet:

"Cyber Essential Scheme Data" sensitive and personal information and other relevant information as referred to in the Cyber Essentials Scheme; and

"Cyber Essentials Plus Certificate" the certification awarded on the basis of
external testing by an independent
certification body of the Contractor's
cyber security approach under the

Cyber Essentials Scheme and is a more advanced level of assurance.

2. CYBER ESSENTIALS OBLIGATIONS

- 2.1 Where the Scope requires that the *Contractor* provide a Cyber Essentials Certificate prior to the execution of the *works* the *Contractor* shall provide a valid Cyber Essentials Certificate, then on or prior to the commencement of the *works* the *Contractor* delivers to the *Client* evidence of the same. Where the *Contractor* fails to comply with this paragraph it shall be prohibited from commencing the carrying out of the *works* under any contract until such time as the *Contractor* has evidenced to the *Client* its compliance with this paragraph 2.1.
- 2.2 Where the *Contractor* continues to Process Cyber Essentials Scheme Data during the carrying out of the *works* the *Contractor* delivers to the *Client* evidence of renewal of the Cyber Essentials Certificate on each anniversary of the first applicable certificate obtained by the *Contractor* under paragraph 2.1.
- 2.3 Where the *Contractor* is due to Process Cyber Essentials Scheme Data after the commencement of the *works* but before completion of the *works* the *Contractor* delivers to the *Client* evidence of:
- 2.3.1 a valid and current Cyber Essentials Certificate before the *Contractor* Processes any such Cyber Essentials Scheme Data; and

- 2.3.2 renewal of the valid Cyber Essentials Certificate on each anniversary of the first Cyber Essentials Scheme certificate obtained by the *Contractor* under paragraph 2.1.
- 2.4 In the event that the *Contractor* fails to comply with paragraphs 2.2 or 2.3 (as applicable), the *Client* reserves the right to terminate this contract for material Default.
- 2.5 The *Contractor* ensures that all sub-contracts with Sub-Contractors who Process Cyber Essentials Data contain provisions no less onerous on the Sub-Contractors than those imposed on the *Contractor* under this contract in respect of the Cyber Essentials Scheme under paragraph 2.1 of this Schedule
- 2.6 This Schedule shall survive termination or expiry of this contract.

Appendix 7

<u>Delivery Contractors – Lot C England, South</u>

REDACTED
REDACTED
REDACTED
REDACTED
REDACTED

Appendix 8

Performance KPIs

PART A

1 Performance Levels

- 1.1 The objectives of the Performance KPIs and Performance Failure Points are to:
 - 1.1.1 ensure that the Works carried out under a Task Order meet the requirements of the Client and that each Task Order is administered in accordance with the provisions of that Task Order;
 - 1.1.2 provide a mechanism whereby the Client can attain meaningful recognition of inconvenience resulting from the Contractor's failure to deliver the Works⁹ for which it has contracted to deliver and/or administer the Task Order in accordance with the provisions of that Task Order; and
 - 1.1.3 incentivise the Contractor to meet and exceed the requirements set out in a Task Order and to remedy any Performance Failure expeditiously.
- 1.2 Without prejudice to the Client's other rights and remedies under this Contract and/or any Task Order, if the level of performance of the Contractor of any element of a Task Order during a Service Period:
 - 1.2.1 achieves the Performance Level in respect of each Performance KPI, no Performance Failure Points will accrue to the Contractor in respect of that Performance KPI; or
 - 1.2.2 is below the Performance Level in respect of each Performance (i.e. a Performance Failure arises), the appropriate number of Performance Failure Points will accrue to the Contractor in respect of that Performance KPI.
- 1.3 If a Performance Failure occurs for two (2) or more Task Orders, the second such failure shall be a "**Repeat Failure**". Any subsequent failure by the Contractor to achieve that same Performance Level shall also be a Repeat Failure.
- 1.4 If any Repeat Failure occurs, the number of Performance Failure Points that shall accrue to the Contractor in respect of such Repeat Failure shall be the number of Performance Failure Points that would normally accrue in respect of an initial failure of that Performance Level multiplied in accordance with the following:
 - 1.4.1 initial failure: 1;
 - 1.4.2 1st Repeat Failure: 1.5;
 - 1.4.3 2nd Repeat Failure: 2.0; and
 - 1.4.4 3rd and subsequent Repeat Failures: 2.5.

2 <u>Performance Failure Points</u>

2.1 Subject to paragraph 3.2, in any Service Quarter, where the Contractor accrues 100 or more Performance Failure Points, the Contractor shall receive a YELLOW Score for the purposes of the Contract.

- 2.2 Subject to paragraph 3.2, in any Service Quarter, where the Contractor accrues 150 or more Performance Failure Points, the Contractor shall receive a RED Score for the purposes of the Contract.
- 2.3 Where the Contractor receives a YELLOW or RED Score, without limitation to the Client's other rights and remedies which are available to it under this Contract, the Client may suspend and/or exclude (as applicable) the Contractor from participating in the Identification Procedure as set out in Part A of Appendix 1.

3 Earn Back Points

- 3.1 Subject to paragraph 4, in any Service Period, where the Contractor meets and/or exceeds the relevant Performance Level, the Contractor shall receive Earn Back Points for the purposes of the Contract.
- 3.2 At the end of each Service Quarter, any Earn Back Points shall be taken into account when determining the amount of Performance Failure Points accrued by the Contractor for any Task Order completed by the Contractor in any Service Quarter.

4 Performance Monitoring Reports

- 4.1 Within twenty (20) Business Days of the end of each Service Quarter, the Contractor shall provide a Performance Monitoring Report to the Client. The Performance Monitoring Report shall be in the format agreed between the Parties and shall contain, as a minimum, the following information in relation to each Task Order in respect of the Service Quarter just ended:
 - 4.1.1 a scorecard summarising the Contractor's level of performance in relation to each Performance Level and setting out the detail in relation to the underlying figures;
 - 4.1.2 a summary of all Performance Failures that occurred during the Service Quarter;
 - 4.1.3 which Performance Failures remain outstanding and progress in resolving them;
 - 4.1.4 for any Performance Failure, the cause of the fault and any action being taken to reduce the likelihood of recurrence;
 - 4.1.5 for any Repeat Failures, actions taken to resolve the underlying cause and prevent recurrence;
 - 4.1.6 the number of Performance Failure Points awarded to each Performance Failure in that Service Quarter:

- 4.1.7 any Earn Back Points that the Contractor considers should be taken into account as part of ongoing performing monitoring;
- 4.1.8 a rolling total of the number of Performance Failure Points that have occurred; and
- 4.1.9 such other details as the Client may reasonably require from time to time.
- 4.2 The draft Performance Monitoring Report shall be reviewed and its contents agreed by the parties at the Performance Review Meeting which immediately follows the issue of such report.
- 4.3 The parties shall attend Performance Review Meetings in each Service Quarter (unless otherwise agreed). The Performance Review Meetings will be the forum for the review by the Contractor and the Client of the Performance Monitoring Reports. The Performance Review Meetings shall take place at a date and time agreed by the Contractor and Client in the Service Quarter following the Service Quarter to which the Performance Monitoring Report pertains and shall be fully minuted.
- 4.4 The Contractor shall provide to the Client such supporting documentation as the Client may reasonably require in order to verify the level of the performance by the Contractor and the calculations of the amount of Performance Failure Points for any specified period.
- 4.5 If the Contractor fails to provide any Performance Monitoring Report in accordance with this paragraph 4 then it shall receive a RED Score.

5 Changes to KPIs

- 5.1 The Client may (acting reasonably):
 - 5.1.1 change the weighting of one or more of Performance KPIs; and/or 5.1.2 change one or more of the Performance KPIs,

and the Contractor shall not be entitled to object to, or increase the Prices, as a result of such changes provided that:

- (i) the Client gives the Contractor at least one (1) Service Quarter's notice (or such shorter notice period as may be agreed with the Contractor); and
- (ii) it is reasonable to make such changes, including (without limitation) where:
 - (A) it becomes apparent that a Performance KPI is failing to accurately capture the required output; and/or
 - (B) where it is necessary to reflect changes in the Client's business requirements and/or priorities; and/or
 - (C) where it is necessary to reflect changing industry standards.

5.2 Where a Performance KPIs is changed in accordance with paragraph 5.1, the change shall be effective from the expiry of the notice referred to in paragraph 5.1, or such later date as the parties may agree.

PART B

120 can use -40 from previous task order above in same quarter to avoid a 'yellow score' acti

Appendix 9

Parent Company Guarantee

Project specific basis as & when required

Performance Bond

Project specific basis as & when required

Client's Schedules

Part A: Appendix G (Collateral Warranty)

Project specific basis as & when required

Part B: Appendix K (TUPE)

Project specific basis as & when required

12

Scope of Requirements

Project specific basis

REDACTED